

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hiesinger, Karolin

Working Paper To include or not to include? Firm employment decisions with respect to the German disabled worker quota

IAB-Discussion Paper, No. 25/2022

Provided in Cooperation with: Institute for Employment Research (IAB)

Suggested Citation: Hiesinger, Karolin (2022) : To include or not to include? Firm employment decisions with respect to the German disabled worker quota, IAB-Discussion Paper, No. 25/2022, Institut für Arbeitsmarkt- und Berufsforschung (IAB), Nürnberg, https://doi.org/10.48720/IAB.DP.2225

This Version is available at: https://hdl.handle.net/10419/270316

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



https://creativecommons.org/licenses/by-sa/4.0/deed.de







IAB-DISCUSSION PAPER

Articles on labour market issues

25|2022 To Include or Not to Include? Firm Employment Decisions with Respect to the German Disabled Worker Quota

Karolin Hiesinger



To Include or Not to Include? Firm Employment Decisions with Respect to the German Disabled Worker Quota

Karolin Hiesinger (Institute for Employment Research (IAB), Germany)

Mit der Reihe "IAB-Discussion Paper" will das Forschungsinstitut der Bundesagentur für Arbeit den Dialog mit der externen Wissenschaft intensivieren. Durch die rasche Verbreitung von Forschungsergebnissen über das Internet soll noch vor Drucklegung Kritik angeregt und Qualität gesichert werden.

The "IAB-Discussion Paper" is published by the research institute of the German Federal Employment Agency in order to intensify the dialogue with the scientific community. The prompt publication of the latest research results via the internet intends to stimulate criticism and to ensure research quality at an early stage before printing.

Contents

1	Introduction6				
2	The	erman Institutional Background9			
	2.1	The Situation for Disabled Workers			
	2.2	The German Disabled Worker Law10			
3	Theo	etical Considerations 12			
	3.1	Employment Decisions at the Quota Thresholds12			
	3.2	Manipulation of Employment at the Second Threshold			
4	Emp	ical Strategy, Data and Variables18			
	4.1	Data18			
	4.2	Empirical Strategy19			
	4.3	Sample and Descriptive Statistics			
5	Resu	ts: Intended and Unintended Effects 24			
	5.1	Demand for Disabled: Graphical Illustration24			
	5.2	Demand for Disabled: Naive Effects			
	5.3	Unintended Effect: Bunching Below			
	5.4	Bounding the Effect			
	5.5	Bunching Behavior			
	5.6	Robustness Checks: Placebos and Donuts31			
		5.6.1 Further Robustness Tests 35			
	5.7	Heterogeneous Effects			
		5.7.1 Low-Wage and High-Wage Firms			
		5.7.2 Effects by Industry			
	5.8	Results for the 60-Employee Threshold 40			
6	Sum	nary and Conclusions 42			
Ap	pend	43			
	Арре	ndix A: Definitions and Institutional Details			
	Арре	ndix B: Further Analyses			
Re	feren	es			

Abstract

This paper analyzes whether financial disincentives affect firm demand for disabled workers. In Germany, firms must pay a noncompliance fine if they do not meet their legal quota for disabled workers. I exploit a threshold in this quota: Firms with fewer than 40 employees are required to employ one disabled worker, whereas firms with 40 or more employees must employ two disabled workers. Using administrative firm data, my results suggest that firms respond partially to the threshold and employ 0.388 more disabled workers when they are located just above the threshold. The effect remains positive after correcting for bunching behavior.

Zusammenfassung

In Deutschland müssen Unternehmen eine Ausgleichsabgabe zahlen, wenn sie die gesetzliche Quote zur Beschäftigung von Menschen mit Schwerbehinderungen nicht erfüllen. Im vorliegenden Papier wird untersucht, inwieweit die Ausgleichsabgabe die Arbeitsnachfrage von Unternehmen beeinflusst. Dabei nutze ich eine Schwellenwertregelung innerhalb der Schwerbehindertenquote: Unternehmen mit mindestens 20, aber weniger als 40 Beschäftigte müssen mindestens eine Person mit Schwerbehinderung beschäftigen, Unternehmen mit mindestens 40, aber weniger als 60 Beschäftigte müssen mindestens zwei Menschen mit Schwerbehinderungen beschäftigen. Mit Hilfe administrativer Unternehmensdaten schätze ich den Schwellenwerteffekt auf die Anzahl der Personen mit Schwerbehinderungen im Unternehmen. Meine Ergebnisse zeigen, dass Unternehmen zum Teil auf die Regelung reagieren und im Durchschnitt 0,388 mehr Personen mit Schwerbehinderungen beschäftigen, wenn sie sich knapp oberhalb des Schwellenwertes befinden. Dieser Effekt bleibt auch dann positiv, wenn berücksichtigt wird, dass manche Unternehmen bewusst unterhalb der Schwelle bleiben.

JEL

J15, J21, J23, J71, J78

Keywords

disability, employment quota, noncompliance fine, administrative data

Acknowledements

I am particularly grateful to Jacopo Bassetto, Mario Bossler, Matthias Collischon, Bernd Fitzenberger, Sofía Fernández Guerrico, Nicole Gürtzgen, Alexander Kubis, Benjamin Küfner, Johannes Ludsteck, Markus Nagler, Laura Pohlan, Martin Popp, Malte Sandner, Claus Schnabel, Gesine Stephan and Rocío Titiunik for helpful discussions and suggestions. I further thank Franka Vetter for excellent research assistance. Earlier versions of this paper were presented at the Annual ESPAnet Conference, the Annual Conference of the Italian Association of Labour Economists, the IAAEU Workshop on Labour Economics as well as at seminars at the IAB (Nuremberg), the IFAU (Uppsala) and the University of Regensburg. I would like to thank all participants for their helpful comments.

Availability of data

In this paper I use proprietary data that is held by the Institute of Employment Research (IAB). Since this data is of administrative origin, access is subject to legal restrictions under the laws of the Ger-man Social Code. Therefore, the data is not freely available to the research community. However, all the datasets and codes of this paper will be archived at IAB exclusively for the purpose of replication.

Funding

The author received no financial support for the research, authorship, and/or publication of this article.

Conflict of Interest

None declared.

1 Introduction

Low levels of employment and high rates of unemployment reflect the considerable labor market disadvantages of individuals with disabilities in all OECD countries. In the late 2000s, the unemployment rate among individuals with disabilities was twice as high as that among individuals without disabilities (14% compared to 7%) (OECD, 2010). Furthermore, many employers prefer to pay a fine instead of employing individuals with (severe) disabilities. In Germany, approximately 27 percent of private employers with 20 or more employees preferred paying a noncompliance fine to employing any worker with a disability in 2019 (Federal Employment Agency, 2021). In addition to discrimination tendencies and prejudices, employers may anticipate higher costs when considering hiring an individual with a (severe) disability. Such individuals may need special workplace equipment, are often subject to special employment protection regulations, take more vacation time and, on average, have higher rates of absence due to illness (Hiesinger/Kubis, 2022).

To improve the integration of disabled individuals into the labor market despite these costs, many OECD countries have implemented policy reforms, often in the form of a mandatory employment quota combined with a monetary fine in the case of noncompliance. Even though employment quotas and noncompliance fines are widely used policy instruments for integrating individuals with severe disabilities into the labor market, surprisingly little is known about their effectiveness.

This paper attempts to analyze the intended and unintended effects of the employment quota for disabled workers. I exploit a threshold in the German labor law that sets the mandatory employment quota: Employees that are below the 40-employee threshold but have at least 20 employees are obliged to employ at least one disabled individual. Above this threshold, employers must employ at least two disabled individuals. My empirical analysis consists of two parts. First, I analyze whether employers (i.e., firms) manipulate their employment levels and purposely stay – or bunch – below the threshold to avoid the fine. I refer to this as the *unintended* effect of the quota. Second, I estimate the *intended* effect of the quota, which is the effect of the threshold on the number of disabled workers in a firm. For this, I follow Lalive/Wuellrich/Zweimüller (2013) and adopt a threshold design which is closely related to a regression discontinuity design (RDD). However, as I do find evidence of bunching, the naive threshold effect is potentially biased. Quantifying the magnitude of the bunching effect helps me to assess this bias and to bound the threshold effect.

Understanding the intended and unintended effects of any employment quota is crucial for two reasons. First, along with antidiscrimination legislation, mandatory employment

quotas are one of the most common policies for integrating disabled workers into the labor market, as they are used in many OECD countries (OECD, 2003, 2010). While the effects of antidiscrimination policies have been quite well explored, there is a remarkable paucity on the effects of employment quotas on firm demand for disabled workers. Given the alleged importance of these quotas for integration, this paucity is striking. Second, my study helps to better understand the role of employment quotas and financial disincentives for labor demand in general. The threshold in the regulatory employment quota implies a sharp change in relative labor costs for different firms near the threshold. Thus, this policy allows me to study the behavior of firms facing this discontinuity. In so doing, I explicitly address the unintended effects of the employment quota, such as adjustments in the (nondisabled) workforce composition or changes in firm dynamics near the threshold. I further differentiate between firms that face different costs at the threshold depending on the extent of their compliance with the quota regulation.

To date, few studies have addressed the impact of an employment quota on firm dynamics and on firm demand for disabled workers. A large number of studies have looked at either the effects of antidiscrimination legislation with respect to workers with disabilities (see, for example, Acemoglu/Angrist, 2001; Beegle/Stock, 2003) or the impact of disability policies on the employment of disabled workers from a labor supply perspective (see, for example, Verick (2004) and Lechner/Vazquez-Alvarez (2011) for Germany or Barnay et al. (2019) for France). However, to the best of my knowledge, only three studies have evaluated the effect of a disabled worker quota on employment decisions from a labor demand perspective.

Lalive/Wuellrich/Zweimüller (2013) examine whether there is a discontinuity in the employment of disabled individuals between firms below and above the Austrian employment quota, which kicks in when a firm reaches 25 nondisabled workers. The authors find that firms react to the quota in two ways. First, firm demand for disabled workers increases above the threshold. Second, some firms manipulate their employment of nondisabled workers and purposely stay below the threshold. Despite this manipulation, the lower bound of the threshold effect remains positive. Similarly, Wagner/Schnabel/Kölling (2001) and Koller/Schnabel/Wagner (2006) examine firm dynamics at quota thresholds in Germany. While Wagner/Schnabel/Kölling (2001) do not find any evidence that there is an effect on employment growth at the first threshold within the employment quota, Koller/Schnabel/Wagner (2006) find evidence that employment growth slows slightly just before the second threshold. Wagner/Schnabel/Kölling (2001) argue that according to their results, the (first) threshold in the German disabled worker law "does not seem to have the kind of strong negative influence on job dynamics in small firms that is often attributed to it in public debates" (p. 10).

I extend this scarce literature on employment quotas and labor demand and study the German case in more detail. I contribute to the literature in two ways: *First*, I revisit the

findings obtained by Wagner/Schnabel/Kölling (2001) and Koller/Schnabel/Wagner (2006) and analyze the German employment quota with a high-quality data set that has more precise information on firm size according to the disabled worker law and the number of disabled workers in a firm. As both Wagner/Schnabel/Kölling (2001) and Koller/Schnabel/Wagner (2006) use establishment-level survey data – and combined with administrative data in the case of Koller/Schnabel/Wagner (2006) – , these studies are based on only a small number of observations (approximately 300-400 establishments). Due to data restrictions, Wagner/Schnabel/Kölling (2001) do not have information on the number of workers with disabilities in each establishment. Furthermore, that study could only approximate the number of workers in each establishment on the basis of the disabled worker law and thus probably suffers from measurement error.

In contrast, the data used in the present study – the Official Employment Statistics on Severely Disabled People – are based on the notification procedure used by the German Federal Employment Agency to determine compliance with the employment quota. Thus, this data set contains firm size information that is consistent with the definition of firm size stipulated in the German disabled worker law.¹ The data set contains information on *all* German firms subject to the employment obligation (i.e., firms with 20 or more employees). Thus, my analyses are based on a vast number of observations. Combined with an additional administrative data set from the Federal Employment Agency, namely, the Establishment History Panel (BHP), I am able to describe all German firms around the firm size thresholds and each firm's workforce in great detail.

Second, while being closely related to Lalive/Wuellrich/Zweimüller (2013), I shed more light on the bunching behavior of firms below the threshold and investigate whether firms adjust their (nondisabled) employment in the face of the threshold. As labor costs increase at the threshold, firms just below the threshold may avoid crossing it. This may be done, for example, by extending the number of hours worked per employee or substituting workers who are counted toward the threshold number of employees with workers who are not counted (e.g., marginally employed workers). While Lalive/Wuellrich/Zweimüller (2013) find that firms below and above the threshold are quite similar in the Austrian case, my results for the German case suggest considerable differences between those firms with regard to firm dynamics, workforce and productivity. I further systematize the potential bunching of firms along the costs these firms face at the threshold. In so doing, I differentiate between *noncompliers*, which face the highest costs at the threshold, *perfect compliers*, which face lower costs at the threshold than *noncompliers*, and *overcompliers*, which do not face any additional costs at the threshold. Analyzing the extent to which these different types of firms bunch helps to clarify the role of (additional) labor costs.

¹ Note that the definition of firm/establishment size in German labor law is not consistent with that used in the German disabled worker law. Depending on the law, the (i) reference point (e.g., establishment, firm or employer), (ii) focal employee group (e.g., freelancers, marginally employed workers or apprentices) and (iii) measure of the number of employees (e.g., per capita or full-time equivalents) differ considerably.

As a preview of my results, I find that firms above the threshold do in fact employ more disabled workers than firms below the threshold. Furthermore, I find clear evidence of firms bunching just below the threshold. Firms purposely stay below the threshold and adjust their workforce accordingly to avoid the (increase in the) noncompliance fine. This bunching is particularly pronounced among *noncompliers*, i.e., those firms that face the largest increase in costs at the threshold. Taking this bunching into account, I assess the bias in the threshold effect and find that even though firms manipulate their employment, the lower bound on the threshold effect is still positive.

The remainder of the paper is structured as follows: Section 2 describes the German institutional setting, and Section 3 discusses the theoretical framework developed by Lalive/Wuellrich/Zweimüller (2013) for the German context. Section 4 presents the data set and the empirical strategy. Section 5 provides the empirical results for the intended and unintended effects, and Section 6 concludes.

2 The German Institutional Background

2.1 The Situation for Disabled Workers

In Germany, a special independent institution (*Versorgungsamt*) grants disability status once a medical expert diagnoses a physical, mental or psychological disorder that is not typical for the age of the patient. This disorder needs to be expected to last longer than six months and needs to impair the ability of the individual to participate in social life. Depending on the extent of the impairment, the medical expert evaluates the degree of the disability ranging from 20 to 100, graduated in steps of ten. An individual is defined as "severely disabled" if his or her degree of disability is greater than or equal to 50.² In the labor market, individuals with a degree of disability between 30 and 50 can be treated as severely disabled when their disability restricts their opportunities to find or hold a job. The decision to obtain disability status and to report that status to an employer is voluntary.

In 2011, approximately 7.3 million individuals (8.9 per cent of the total population) in Germany were considered severely disabled. Since then, the number has continued to increase to over 7.9 million in 2019 (9.5 percent). Data from the Federal Statistical Office

² An example of a disability of degree 50 is voicelessness or a lip-jaw cleft until closure of the jaw cleft.

from 2011 show that disabilities occur mainly in older people. A total of 53.4 percent of severely disabled individuals in Germany in 2011 were 65 years or older. The vast majority of disabilities – approximately 85 percent – are caused by illness. Hence, only a small percentage of disabilities are congenital or due to war injuries, accidents or other causes.

With regard to the degree of disability, almost a quarter (24.3 percent) of severely disabled individuals were assigned the highest degree of disability (100) in 2011, while 31.4 percent had a degree of disability of 50. Physical causes – in particular organ disorders – account for the majority of disabilities (approximately 62.3 percent). A total of 11.1 percent of disabled individuals had mental or emotional disabilities, and 9.0 percent suffered from cerebral disorders. For the remaining fraction (17.6 percent), the type of the most severe disability is not indicated (Federal Statistical Office, 2013).

2.2 The German Disabled Worker Law

The legal framework for promoting the integration of people with disabilities in the labor market in Germany is laid down in part 3 of Book IX of the Social Code, "Integration and Rehabilitation of Disabled People (SGB IX, 2001)", also called the disabled worker law (*Schwerbehindertenrecht*). Enacted in 2001, it built upon the People with Severe Disabilities Act (PSDA), which was originally implemented in 1974. In 2018, the *Bundesteilhabegesetz* replaced the former law.³ One key element of the disability law is the *employment obligation* for public and private employers to fill at least 5 percent of their positions with severely disabled workers. Many other OECD countries, such as Austria, France, Italy and Spain, use similar quota systems to enforce the employment of workers with severe disabilities (OECD, 2003, 2010).

Key to my analysis is the fact that the quota system applies only to employers exceeding a stipulated size.⁴ Small firms with fewer than 20 (nondisabled and disabled) employees are exempt from the employment obligation.⁵ Firms with 20 to less than 40 employees must employ at least one severely disabled individual, whereas firms with 40 to less than 60 employees must employ at least two disabled individuals. Firms with 60 or more employees must meet the 5 percent quota. Firms that do not comply with this obligation must pay a

³ Between 2001 and 2018, some marginal changes were made to the law. These changes include, for example, the role of the Federal Employment Agency in integrating disabled individuals into the labor market. However, these changes do not relate to the employment obligation or the noncompliance fine and therefore should not affect the empirical analysis.

⁴ "Employers" can be natural persons or legal entities under public or private law. To measure employer size, all employees of the same employer are counted together, regardless the number of establishments or other workplaces across which they are distributed (see also Koller/Schnabel/Wagner, 2006). Thus, in the following, "employer" is used synonymously with "firm".

⁵ The German regulation refers to total employment in a firm, including disabled and nondisabled workers. In contrast, the Austrian quota, for example, refers only to the nondisabled workers.

graduated noncompliance fine (*Ausgleichsabgabe*).⁶ Figure 1 provides an overview of the German quota regulation and the corresponding noncompliance fines. The purpose of this noncompliance fine is to compensate firms that fulfil the employment obligation for costs incurred.⁷ Such costs may arise, for example, from the need to purchase special workplace equipment for the disabled worker. Further costs may arise as employees with a recognized status "severely disabled" status are institutionally better protected in two ways. First, they are subject to special dismissal protection. If the employee has been working longer than six months in a firm, the employer needs to obtain permission for a dismissal from the local integration office. Second, a severely disabled worker receives more vacation days, i.e., an additional five days per year.



Figure 1: Employment Obligation and Monthly Noncompliance Fines in Germany

Notes: The figure shows the legal regulations concerning the German employment quota and noncompliance fines (NCF) according to § 159 SGB IX during the observation period (2004–2011). L^D is the number of workers with disabilities that firms are obligated to employ; $L^N + L^D$ represents the number of employees in a firm (i.e., firm size). For details on the firm size calculation, see Table A.1 in the Appendix. The noncompliance fines increased in 2012, 2016 and 2021. The current fines are 140, 245 and 360 EUR/month, respectively. *Source:* Own illustration.

As in almost all countries with a quota system, the employment quota is generally not met in Germany. In 2011, 60.1 percent of employers with 20 or more employees did not fulfill the employment obligation and thus had to pay the noncompliance fine. Furthermore, approximately one quarter (26.2 percent) did not employ any severely disabled workers. These percentages have remaind essentially unchanged since then. In general, public

⁶ Note that paying the noncompliance fine does not remove the employment obligation. Thus, employers can be fined up to 10,000 EUR in addition to the noncompliance fine if they culpably fail to comply with the employment obligation (§ 238 SGB IX). However, these fines are rarely imposed. For example, in 2010, only two fines were imposed (German Bundestag, 2011).

⁷ The fine must be paid to the integration offices and is used mainly to finance assistance for occupational rehabilitation for severely disabled individuals. In 2019, the revenue from the noncompliance fine in Germany amounted to almost 696 mil. EUR (Bundesarbeitsgemeinschaft der Integrationsämter und Hauptfürsorgestellen, 2020).

employers are better at fulfilling their quotas. The share of workers with a disability is particularly low in the hotel and restaurant industry and in the agricultural sector (Federal Employment Agency, 2014).

3 Theoretical Considerations

This section mainly discusses the behavioral framework developed by Lalive/Wuellrich/Zweimüller (2013). The basic idea behind this framework is that a threshold determining the quota for workers with disabilities may affect the demand not only for disabled workers but also for nondisabled workers. In what follows, I reformulate this framework for the German quota system. This framework serves to explain the bunching behavior of firms below the quota threshold T, where T refers to the total number of workers in the firm. To examine this behavior, I look at firms with T - 1 employees and their decision to hire an additional (disabled or nondisabled) worker.⁸

3.1 Employment Decisions at the Quota Thresholds

Lalive/Wuellrich/Zweimüller (2013) assume that nondisabled workers have productivity P, whereas disabled workers have productivity p which is less than P.⁹ Because of antidiscrimination legislation, both disabled and nondisabled workers receive the same wage w.¹⁰ The productivity of nondisabled workers exceeds the wage (P > w) in all firms, but firms differ in the value of p that obtains, i.e. p < w in some firms and p > w in other firms.¹¹ As the quota rule is based on a head count, labor L is assumed to be indivisible, while product demand Z is assumed to be continuous. Labor consists of L^N nondisabled and L^D disabled workers.

⁸ Note that the framework presented below refers to the *first* threshold in Germany (20 employees) in order to explain the general logic behind the quota system. This logic also generalizes to higher thresholds. In subsection 3.2, I also discuss the implications of the framework for employment manipulation at the *second* threshold (40 employees).

⁹ Assuming that p < P is plausible in the German context: Survey results show that a considerable share of German establishments report a lower level of performance and resilience and a higher level of absence rates among workers with disabilities than among workers without disabilities (Hiesinger/Kubis, 2022).

¹⁰ Note that I assume that firms face the same hiring costs for disabled or nondisabled workers. But as disabled individuals may have a lower labor force participation, hiring disabled workers may result in higher search costs. These additional search costs would then increase the marginal costs of hiring disabled workers. However, for the sake of simplicity, I abstract from including these additional costs in my framework.

¹¹ Note that a workers' productivity within a firm may decline depending on the level of employment (see also footnote 15).

Let us first discuss this system in the *absence of a quota*. The firm's profit function can be described as

$$\pi_0(L^N, L^D) = \min(PL^N + pL^D, Z) - w(L^N + L^D)$$
(1)

"Residual demand" is defined as the product demand Z minus the output produced by the L^N nondisabled workers:

$$R(Z, L^N) = Z - L^N P \tag{2}$$

Figure 2 illustrates the firm's employment decisions in the absence of a quota for p < w and p > w. In either case, the firm will not hire an additional nondisabled worker ($\Delta L^N = 0$) as long as residual demand is below the wage rate: $R(Z, L^N) < w$ (A- and D-firms). When residual demand exceeds the wage rate ($R(Z, L^N) > w$), the firm will hire an additional nondisabled worker ($\Delta L^N = 1$) (C-firms). However, the firm is not willing to hire a disabled worker ($\Delta L^D = 0$) as long as p < w: The productivity of the disabled worker is always too low to satisfy the residual demand. When p > w and residual demand is in the range $R(Z, L^N) \in (w, p)$, the firm is indifferent between hiring a disabled or a nondisabled worker, as the productivity of both workers is great enough to satisfy the residual demand (*F*-firms). A firm with large residual demand $p < R(Z, L^N) < P$ will strictly prefer hiring a nondisabled worker over a disabled worker (*H*-firms).

Now let us describe a system in the *presence of a quota*. In Germany, firms with total employment below the threshold T (i.e., $L^N + L^D < T$) do not face an employment obligation for disabled workers, whereas firms with total employment at or above T (i.e., $L^N + L^D \ge T$) need to employ at least one disabled worker. Note that the German system differs from the Austrian system presented in Lalive/Wuellrich/Zweimüller (2013). Specifically, the quota in Germany is determined on the basis of total employment, i.e., nondisabled *and* disabled employment ($L^N + L^D$), whereas the quota system in Austria is based only on nondisabled (L^N) employment. Firms must pay a noncompliance fine τ if they do not satisfy their employment obligation. Following Lalive/Wuellrich/Zweimüller (2013), I assume that $\tau < w$ and $\tau < P - p$.¹² The profit function in the presence of a quota given nondisabled employment L^N and disabled employment $L^D \in \{0, 1\}$ can be

¹² Both assumptions are plausible in the German context. First, among firms with 20-60 employees, the noncompliance fine is only approximately 4.3-7.5% of gross monthly earnings. Second, the degree of disability (see Section 2.1) reflects the extent of the impairment caused by the disability. For severely disabled workers (those with a degree of disability of at least 50), it is plausible to assume that this impairment substantially affects their labor productivity. Furthermore, survey results show that a considerable share of German establishments report that workers with disabilities have a lower level of performance and resilience and a higher rate of absence than workers without disabilities (Hiesinger/Kubis, 2022).

Figure 2: No Quota



Notes: The figure shows the employment decisions of firms with T - 1 employees in the absence of a quota system.

Source: Own illustration based on the discussion in Lalive/Wuellrich/Zweimüller (2013).

described as

$$\pi_1(L^N, L^D) = \min(PL^N + pL^D, Z) - w(L^N + L^D) + \min[L^D - 1(L^N + L^D \ge T), 0] \cdot \tau$$
(3)

with
$$(L^N + L^D \ge T) = \begin{cases} 1 & \text{if } L^N + L^D \ge T \\ 0 & \text{if } L^N + L^D < T \end{cases}$$

For employment decisions in the presence of a quota, I again distinguish between the cases p < w and p > w, as shown in Figure 3. When p < w, firms will not hire any disabled workers, even in the presence of a quota ($\Delta L^D = 0$). However, the quota may affect nondisabled employment. A firm with residual demand R in the range $(w, w + \tau)$ will not hire an additional nondisabled worker ($\Delta L^N = 0$), whereas it would have hired that worker in the absence of the quota (see Figure 2). The marginal cost an additional nondisabled worker is now the wage w of this worker plus the tax τ . In the range $(w, w + \tau)$, this marginal cost is larger than the residual demand. Thus, firms with residual demand in this range, i.e., *B-Firms*, are better off setting their employment level just below the threshold, i.e.,

 $L^N + L^D = T - 1$, to avoid the tax. Avoiding the tax by staying below the threshold – i.e., bunching – is the unintended effect of the quota.

For p > w, the decision to hire a disabled worker depends on the residual demand R. The quota rule does not affect *D-firms*, which have a low level of residual demand in the range (0, w). As residual demand does not exceed the wage rate, an additional worker – disabled or not – would not produce the needed revenue. However, firms with residual demand in the range (w, p), *F-firms*, and in the range $(p, p + \tau)$, *G-firms*, now prefer to hire a disabled worker instead of a nondisabled worker as they would have to pay the (additional) fine when hiring a nondisabled worker (without a quota, *F-firms* are indifferent between hiring a disabled workers). Incentivizing firms to hire a disabled worker instead of a nondisabled worker while *G-firms* prefer to hire a nondisabled worker and no disabled workers). Incentivizing firms to hire a disabled worker instead of a nondisabled worker while *G-firms* prefer to hire a nondisabled worker and no disabled workers). Incentivizing firms to hire a disabled worker instead of a nondisabled worker set higher profits despite the fine.¹³

Let us now have a closer look at manipulating firms. Lalive/Wuellrich/Zweimüller (2013) define *manipulators* as firms that set their employment level below the threshold under the quota system but above it without such a system. The Austrian system induces *B-firms* and *G-firms* to manipulate employment, as the Austrian quota is defined on the basis of nondisabled employment. *B-firms* do not hire a disabled worker and set their nondisabled employment level below the threshold to avoid the tax. *G-firms* also set their nondisabled employment level below the threshold but hire a disabled worker (because he or she increases profit). In the Austrian system, *G-firms* that hire a disabled worker are still located *below* the threshold. However, in contrast to the Austrian system, the German system induces only *B-firms* to manipulate their employment. As the German quota is based on total – i.e., nondisabled and disabled – employment, *G-firms* cross the threshold when they hire a disabled worker and are located *above* the threshold. Thus, I expect potential manipulation to arise entirely from *B-firms*, for which p < w and which purposely stay below the threshold to avoid the fine.

How does this manipulation bias the difference in the average number of disabled workers between firms with T - 1 employees and firms with T employees? Due to the quota, the composition of firms with T - 1 employees changes: *B-firms* would have hired an additional nondisabled worker without the quota (i.e., they would have chosen employment level T) but now bunch below the threshold in the presence of the quota. As *B-firms* are not willing

¹³ Lalive/Wuellrich/Zweimüller (2013) also discuss employment decisions for firms at or above the threshold T (where T refers to the number of nondisabled workers), i.e., the decision to hire T or T + 1 workers. Those firms will hire an additional disabled worker when the residual demand is in the range $(w - \tau, w)$, as the marginal cost of hiring a disabled worker is $w - \tau$ and thus less than the residual demand. As I focus on employment decisions *below* the threshold, i.e., the decision to hire T - 1 or T workers, I do not discuss this case in more detail.



Notes: The figure shows the employment decisions of firms with T-1 employees in the presence of a quota system.

Source: Own illustration based on the discussion in Lalive/Wuellrich/Zweimüller (2013).

to hire an additional disabled worker, the difference in the average number of disabled workers between firms below and at the threshold is *overestimated*.

Manipulating firms may further adjust their employment (Koller/Schnabel/Wagner, 2006): To avoid crossing the threshold, *B-firms* may extend the number of hours worked for (incumbent) employees. Note that forcing workers to work overtime could be costly due to overtime pay. However, a firm may, for example, substitute part-time workers with full-time workers. A second option includes substituting workers who are counted when determining whether the firm is subject to the quota (e.g., regularly employed workers) with workers who are not so counted (e.g., marginally employed workers). Note that this would only be the case when the productivity of the members of these working groups is sufficient to meet the product demand. In sum, bunching may lead to different wage and employment structures in a firm.

3.2 Manipulation of Employment at the Second Threshold

As my empirical analysis focuses on the second threshold of 40 employees, let us now briefly discuss the marginal cost for firms with employment just below this second threshold (i.e., $L^N + L^D = T_2 - 1$). As shown in Figure 1, the German labor law defines two levels of the fine depending on the initial level of disabled employment around the second threshold (τ_1 and τ_2 with $\tau_2 > \tau_1$).¹⁴ Thus, there are three types of firms with $L^N + L^D = T_2 - 1$ employees: First, firms with an initial level of disabled employment $L^D = 0$ already pay the noncompliance fine τ_1 . When crossing the threshold, the noncompliance fine increases to $2 \cdot \tau_2 - \tau_1$, provided that it is not the hiring of disabled worker that causes the firms to cross the threshold. I refer to these firms as noncompliers. Second, firms with an initial level of disabled employment $L^D = 1$ are in perfect compliance with the quota rule. When crossing the threshold, these firms would be obliged to pay the noncompliance fine τ_1 , again provided that they do not hire another disabled worker. I refer to these firms as perfect compliers. In the German context during the observation period, perfect compliers (noncompliers) face additional costs of 1260 EUR/year (3060 EUR/year) at the threshold. Finally, firms with $L^D > 2$ already employ more disabled workers than required by law. These firms do not face any additional costs at the threshold, as they do not have to pay a noncompliance fine regardless of whether they are below or above T_2 . I refer to these firms as *overcompliers*.

What are the consequences for employment manipulation at T_2 ? As shown in Section 3.1, employment manipulation arises entirely from *B-firms*, for which p < w. I identify two types of manipulating firms according to the marginal cost that they face at T_2 . First, *B-noncompliers* are firms with no disabled workers and with residual demand in the range $(w, w + 2 \cdot \tau_2 - \tau_1)$. Second, *B-perfect compliers* are firms that already have one disabled worker and residual demand in the range $(w, w + \tau_1)$.¹⁵ Since the range of bunching is larger among *B-noncompliers* as shown in Figure 4, I expect bunching to be more pronounced among this type of firm than among *B-perfect compliers*.

¹⁴ Note that I treat the *initial* level of disabled employment as given and discuss only the decision of firms with $L^N + L^D = T_2 - 1$ employees to hire an *additional* disabled or nondisabled worker.

¹⁵ The fact that I allow *perfect compliers* to bunch even though they have already hired one disabled worker may be rationalized by a decreasing marginal product. This assumption implies that beyond a given level of employment, the ratio of p to w may change. Thus, for some firms, the productivity of the first disabled worker exceeded the wage (i.e., p > w) when the firm had a lower employment level (for example, at the first threshold), while this is no longer the case at the second threshold (i.e., p < w).

Figure 4: Types of B-Firms at T_2



Notes: The figure shows the employment decisions of firms at the first (top panel) and the second threshold (bottom panel). *Perfect compliers* are firms with one disabled worker, and *noncompliers* are firms with no disabled workers. *A-firms* are firms that do not hire a nondisabled worker, whereas *C-firms* are firms that hire a nondisabled worker. *B-firms* are firms that would hire a nondisabled worker in the absence of a quota system but would not hire a nondisabled worker in the presence of such a system (see Figures 2 and 3). *Source:* Own illustration.

4 Empirical Strategy, Data and Variables

4.1 Data

My empirical analysis is based on two administrative data sets from the German Federal Employment Agency. The *Employment Statistics for Severely Disabled People* (BsbM) is an annual set of statistics that has been available since 2003 and that includes information on the employment of disabled workers in firms. Firms with 20 or more employees must annually declare (i) how many individuals they employ and (ii) how many of them are severely disabled. Thus, the information on firm size and the number of disabled workers stems directly from the notifying procedure used to determine compliance with the disabled worker quota. As a consequence, the BsbM has the great advantage of providing information on firm size that is consistent with the legal definition stipulated in the disabled worker law.¹⁶ Note that many studies that have analyzed regulations with a firm size criterion, e.g., in the context of dismissal protection, have tried to approximate the firm size stipulated in the respective law (see, for example, Wagner/Schnabel/Kölling, 2001; Bauer/Bender/Bonin, 2007; Bauernschuster, 2013; Hijzen/Mondauto/Scarpetta, 2017).

¹⁶ For details on the definition of firm size according to the disabled worker law, see Table A.1 in the Appendix.

Thus, these analyses often suffer from a considerable amount of measurement error, which can be ruled out in my case. In addition to some basic information about the firm, such as region and industry, the BsbM contains an identifier for the establishment or the main establishment in the case of multiestablishment firms.

This identifier allows me to merge additional information from the establishment data of the Federal Employment Agency, namely, the *Establishment History Panel* (BHP) (Schmucker et al., 2018). Since I consider only small businesses with up to a maximum of 80 employees, I can assume that in most cases, each firm consists of one establishment.¹⁷ This allows me to merge the firm data with the establishment data. The Establishment History Panel provides annual detailed information on each establishment's workforce, such as its skill or employment structure, as of the reference date, June 30th.

4.2 Empirical Strategy

The employment obligation for firms, which varies according to the firm size thresholds, provides a natural application for a "threshold design" (Lalive/Wuellrich/Zweimüller, 2013).¹⁸ I use the second threshold and compare the various outcome variables, most importantly the number of workers with disabilities, of firms just below and just above the threshold of 40 employees.¹⁹ The key assumption for identifying the effect of the quota is that firm demand for disabled workers would be continuous in the absence of the employment obligation. This assumption is reasonable, as no rules – other than the disabled worker quota – take effect when firms change their employment levels around the thresholds. However, as the noncompliance costs rise sharply at the thresholds, firms may

¹⁷ According to the establishment panel, a representative survey of establishments in Germany, a large majority of establishments are independent companies without any other places of business. This is particularly true for small establishments, thus minimizing the error from treating establishments as single firms. For details, see Figure A.1 in the Appendix. In the case of multiestablishment firms, the establishment information in the *Establishment History Panel* (i.e., the information on the employment, wage and skill structures) refers only to the main establishment. Thus, for multiestablishment firms, the "bunching behavior" analyses (see Section 5.5) would be biased if the wage, skill and employment structures in the main establishment differs substantially from the wage, skill and employment structures in the branch offices. As a robustness check, I exclude those firms that I can identify as multiestablishment firms. For more details, see Section 5.6.1.

¹⁸ Although closely related to an RDD, the threshold design has a slightly different setup than the RDD, as the running variable – firm size in my case – is an endogenous variable. The estimation techniques are, however, very similar.

¹⁹ Due to data limitations, I cannot exploit the first threshold of 20 employees, as the BsbM data set covers only firms affected by the employment obligation, i.e., firms with 20 or more employees. Furthermore, I do not focus on the third threshold of 60 employees for two reasons. First, the assumption that a firm consists of only one establishment (see Section 4.1) is more plausible for smaller firms. Second, there are additional labor law rules with thresholds (apart from the disabled worker quota) for firms with at least 60 employees (see Section 5.8). Thus, I cannot ensure that the effects that I find for this threshold are due solely to the disabled worker quota.

indeed choose to manipulate their employment levels in the presence of the disabled worker quota and purposely stay below the threshold to avoid this additional fine.

Therefore, following Lalive/Wuellrich/Zweimüller (2013), my empirical analysis consists of two parts. First, I estimate the intended threshold effect, which is the (naive) effect of the threshold regulation on the number of disabled workers. Second, I report the unintended bunching effect, which is the effect of the threshold regulation on firm size. The bunching effect thus indicates the maximum number of firms at the threshold that manipulate their size. Taking this potential bunching effect into account, I am able to bound the threshold effect.

To estimate the *threshold effect*, I rely on graphical analyses to provide initial intuition. For this, I plot the local averages of the number of disabled employees per firm size category. In my case, the firm size categories are defined by the whole numbers of employees in a firm. I complement this nonparametric analysis with weighted local polynomial regressions using the following equation:

$$Y_{i} = \beta_{0} + D_{i} \beta_{1} + (1 - D_{i}) c_{i} \beta_{2}^{-} + D_{i} c_{i} \beta_{2}^{+} + (1 - D_{i}) x_{i} \beta_{3}^{-} + D_{i} x_{i} \beta_{3}^{+} + \epsilon_{i},$$
(4)

where Y_i is the outcome variable, i.e., the number of disabled workers in firm *i*. D_i is a treatment dummy indicating whether the firm is above the critical threshold of 40 employees and thus obliged to employ one additional severely disabled worker according to the law. c_i is the running variable that the cutoff is based on (firm size).²⁰ Both c_i and x_i are defined as deviations from the treatment cutoff, i.e., $c_i = C_i - C$, $x_i = X_i - C$, where C denotes the cutoff and X_i represents a vector of control variables capturing predetermined observable firm characteristics.²¹ Including these predetermined characteristics helps to reduce the sampling variability in the estimator. The superscripts '-' and '+' indicate whether the coefficient relates to the left- or right-hand side of the threshold. ϵ_i represents the error term. All coefficient estimates are obtained from a local linear regression that weights all observations by their deviations from the cutoff using a triangular kernel.²²

The aim of the analysis is to extrapolate the counterfactual number of disabled workers in firms at the threshold in the absence of the noncompliance fine. Equation 4 assumes a linear functional form (with polynomial order p=1), which can be misspecified. Thus, to assess the sensitivity of estimates to the functional form, I add higher-order polynomials to

²⁰ Note that this running variable is discrete and takes on 40 distinct values (mass points) between 20 and 59 employees. However, as the number of observations per mass point is sufficiently large (approximately 7,000-30,000 observations per mass point), I can apply the continuity approach presented above (Cattaneo/Irobo/Titiunik, 2018).

²¹ In my case, C_i refers to the whole number of (disabled and nondisabled) employees in firm *i*.

²² This weight is optimal in the MSE-optimal context (Cattaneo/Idrobo/Titiunik, 2019).

the linear model. In so doing, I additionally use polynomials with respect to the running variable of order 2, 3 and 4. Regarding the bandwidth – the window of relevant observations around the threshold – I choose a mean square error optimal (MSE-optimal) bandwidth for each side of the threshold (Calonico/Cattaneo/Farrell, 2020; Cattaneo/Vazquez-Bare, 2016).²³ Furthermore, as I use pooled cross-sectional firm data, I display standard errors adjusted for clustering at the firm level.²⁴

To estimate the unbiased effect of the threshold on the number of disabled workers in the firm, I have to assume that firms do not manipulate their firm size in order to purposely stay below the threshold. However, as (noncomplying) firms face an increase in labor costs at the threshold due to the increased noncompliance fine, this assumption may possibly be violated (see Section 3). Thus, I explicitly address the question of how the manipulation of employment level may bias the estimated naive threshold effect. To do so, I first check whether manipulation is present by graphically inspecting firm size density. The intuition behind this test is that bunching should be reflected as a discontinuity in the firm size distribution at the threshold (see McCrary, 2008). Due to the increased labor costs at the threshold, I expect a negative discontinuity in firm size density at the threshold. I also formally check for the presence of bunching (Cattaneo/Jannson/Ma, 2020). Furthermore, again following Lalive/Wuellrich/Zweimüller (2013), I quantify the effect on the firm size density – the bunching effect – to assess the bias in the estimated naive threshold effect. For this, I use an equation similar to equation (1) but with firm size density (in percentage terms) as the outcome variable. Again, I estimate different specifications, including specifications with different polynomial orders.

To shed additional light on the bunching behavior of firms, I further inspect alternative outcome variables by replacing the dependent variable in equation 4 with each of the alternative outcome variables. These variables include information on firm workforce composition, firm productivity and firm dynamics. Firms just below the threshold that

 $se_{twoway} = \sqrt{se_1^2 + se_2^2 - se_{1\cap 2}^2}$ (see the notes to Table 2).

²³ The form of the MSE-optimal bandwidth is $h_{MSE} = C_{MSE} \cdot n^{-1/(2p+3)}$ (Cattaneo/Vazquez-Bare, 2016). n indicates the sample size available, and p indicates the polynomial order. The constant C_{MSE} involves several known and unknown values that depend on objects such as the kernel function, the parameter of interest, p, the asymptotic bias and variance of the estimator, and whether additional predetermined covariates are included in the estimation. h_{MSE} is constructed by forming a preliminary estimator \hat{C}_{MSE} of the unknown constant C_{MSE} , which leads to the estimated bandwidth $\hat{h}_{MSE} = \hat{C}_{MSE} \cdot n^{-1/2p+3}$. Thus, the selected bandwidth around the threshold T takes the form $[T - \hat{h}_{MSE}, T + \hat{h}_{MSE}]$. As a consequence, only observations within this bandwidth are used. This estimator is data-driven and objective. Note, however, that one cannot directly use the MSE-optimal point estimation. Thus, bias and variance are balanced in a manner that makes inference invalid by construction when the same observations and estimator are used. Calonico/Cattaneo/Titiunik (2014) propose an inference approach based on bias correction for the point estimate. Hence, the *robust* confidence intervals are fully compatible with the use of the observations in the selected MSE-optimal bandwidth and are still valid (Cattaneo/Vazquez-Bare, 2016).

²⁴ Furthermore, following the guide for multiway clustering by Cameron/Miller (2015), I provide standard errors clustered at the firm level and at the discrete values of the running variable (firm size) for my main specification in column (5) of Table 2. To do so, I calculate the standard errors using the following equation:

manipulate their employment levels may substitute regular (full-time) employed workers with workers whose employment does not count toward their firms size for purposes of the quota, such as marginally employed, part-time workers (<18 hours/week) or apprentices. Such substitution effects would be reflected in differences in the workforce composition and in firm productivity below and above the threshold. Another alternative outcome variable is employment growth, as manipulating firms may have lower employment growth just below the threshold.²⁵ As I expect different bunching behavior among different types of firms below the threshold, I always distinguish between *noncompliers*, *perfect compliers* and *overcompliers* (see Section 3).

In summary, my empirical approach explicitly takes potential violations of the key assumptions of a standard RDD/threshold design into account. Specifically, my approach accounts for the fact that observations just below and just above the threshold may indeed be different with regard to workforce composition, productivity and dynamics. However, with regard to *predetermined covariates* such as region, industry and firm age, the observations below and above the threshold should not differ substantially. I first report on these predetermined covariates for firms located around the threshold of 40 employees and formally check for discontinuities at the threshold. Again, I replace the dependent variable in equation 4 with each of the predetermined covariates (see columns (4) and (5) of Table 1). Testing for local balance in the predetermined covariates is important to ensure that firms just below the threshold represent an appropriate control group for treated firms just above the threshold. Furthermore, I include those predetermined covariates as control variables in my main estimations.

4.3 Sample and Descriptive Statistics

As described earlier, I focus on the second threshold of 40 employees in my main analysis. My baseline sample consists of firms with 29 to 51 employees (according to the BsbM) in the years 2004 to 2011, resulting in 319,939 firm-year observations.²⁶

Table 1 reports predetermined firm characteristics (firm age, region and industry) for firms around the threshold of 40 employees. By construction, firms above the threshold have more employees than control firms and are, on average, older. Firms below and above the threshold also differ with regard to their industrial and geographical distribution. However, note that even though all differences are statistically significant at the 1 percent level, most

²⁵ I define employment growth with a dummy variable equal to 1 if a firm has more employees (according to the BsbM) in t + 1 than in t and equal to 0 otherwise.

²⁶ Note that restricting the sample to firms with 29 to 51 employees is only relevant for describing the predetermined characteristics of the firms. In the analysis, the sample differs across the different specifications, as I calculate the MSE-optimal bandwidth for each estimation.

of them are small in size. The mean differences between firms below and above the threshold may also reflect heterogeneity in the firm size distribution across industries and regions. Thus, I formally test for discontinuities in these characteristics at the 40-employee threshold for polynomials of both order 1 and order 4 with an MSE-optimal bandwidth. Columns (4) and (5) of Table 1 report the estimated coefficients. The results for the specification with a very flexible functional form (*p*=4) show that there is only one statistically significant and sufficiently large coefficient (see column (5) of Table 1). In particular, the share of public administration firms is significant (and quite large) discontinuity exactly at the threshold. Hence, I exclude public administration firms in a robustness check to analyze whether my main results are sensitive to this exclusion. Altogether, this inspection of differences in predetermined characteristics suggests that firms below the threshold represent a basically appropriate control group for firms above the threshold.

	Below Threshold	Above Threshold		Dise	continuity
	29-39 Employees	40-51 Employees	Difference	At 40-Employee Thresho	
	(1)	(2)	(3)	(4)	(5)
	Mean	Mean	t test	p = 1	<i>p</i> = 4
Firm Size	33.64	45.18	11.53***		
Age of Establishment	18.80	19.17	0.370***	0.403**	0.954*
Region: East Germany	0.171	0.171	-0.001***	-0.004	0.009
Industry Shares					
Agriculture	0.022	0.016	-0.006***	-0.000	0.003
Energy/Mining	0.009	0.012	0.003***	0.001	0.002
Manufacturing	0.245	0.270	0.025***	0.013*	0.039
Construction	0.096	0.082	-0.014***	-0.007*	-0.029
Wholesale	0.182	0.172	-0.009***	0.005	0.005
Traffic/Communication	0.066	0.062	-0.004***	-0.010***	0.024
Banking/Insurance	0.010	0.014	0.004***	0.007***	0.011*
Other Services	0.188	0.175	-0.013***	-0.021**	-0.021
Public Administration	0.137	0.158	0.021***	0.022***	0.037**
Public Sector	0.045	0.039	-0.005***	-0.002	-0.009
# of Firm-Year Observations	202,583	117,356	319,939		

Table 1: Descriptive Statistics for Firm Characteristics

Notes: The table presents descriptive statistics for the characteristics of firms around the 40-employee threshold. p indicates the order of the polynomial in the specification. *, ** and *** denote statistical significance at the 10%, 5% and 1% levels. The standard errors for the estimates in columns (4) and (5) are clustered at the firm level.

5 Results: Intended and Unintended Effects

5.1 Demand for Disabled: Graphical Illustration

Let us now turn to the graphical illustration of a potential discontinuity at the second threshold of 40 employees. Figure 5 displays the mean number of disabled workers by firm size for the threshold of 40 employees. It shows that the number of disabled workers employed by firms increases with firm size in a quite linear fashion. Firms at the bottom of the observed firm size distribution, i.e., firms with 20 employees, employ on average 0.47 disabled workers, whereas firms at the top of the observed firm size distribution, i.e., firms with 59 employees, employ on average 1.42 disabled workers. The plot shows a considerable discontinuity in the number of disabled workers at the threshold. While firms just below the threshold, i.e., firms with 39 employees, employ on average 0.817 disabled workers, firms just above the threshold, i.e., firms with 40 employees, employ 1.164 disabled workers.

However, the figure also illustrates that the (linear) increase in the number of workers with disabilities slows and eventually reverses into a decline just before the threshold, which can be interpreted as a first indication of bunching behavior: those firms that do not employ enough workers with disabilities may purposely stay below the threshold.

5.2 Demand for Disabled: Naive Effects

Table 2 reports the econometric results for the estimated (naive) threshold effects. I estimate five models with different bandwidths and polynomial orders. The first model in column (1) shows the results for the basic econometric model with an MSE-optimal bandwidth on either side of the threshold, a linear functional form, and predetermined firm characteristics included as control variables. The estimated discontinuity at the threshold is 0.345. This discontinuity is significantly different from zero at the 1 percent level. Columns (2), (3) and (4) use higher-order polynomials and again estimate the optimal bandwidth below and above the threshold. The results show that the estimates are sensitive to functional form. Higher-order polynomials lead to larger threshold effects. This is not surprising, as a more flexible functional form takes the (nonlinear) developments near the threshold into account (see Figure 5). Column (5) also uses a very flexible functional form but with a fixed bandwidth of $h_{below} = 8$ and $h_{above} = 9$ based on the optimal bandwidth in

Figure 5: Number of Disabled Workers



Note: This graph plots the average number of disabled workers by firm size around the 40-employee threshold. The black line approximates the functional form of the running variable (here with polynomial order p = 4 and bandwidth $h_{below} = 20$ and $h_{above} = 19$). *Source:* BsbM and BHP 2004–2011, own calculations.

column (4).²⁷ The estimated coefficients in column (4) and column (5) are very similar. I therefore adopt the model in column (5) with a threshold effect of 0.388 as my *baseline specification* for the remainder of the analysis.

In quantifying the magnitude of the naive effect, the estimates suggest that the employment obligation leads to threshold firms employing 0.388 more disabled workers. Given that the mean number of disabled workers just below the threshold is 0.817, this effect represents an increase in the number of disabled workers of 47 percent. This effect is considerably larger than the 12 percent effect that Lalive/Wuellrich/Zweimüller (2013) found in their analysis of the Austrian case. However, Lalive/Wuellrich/Zweimüller (2013) also found relatively small bunching effects. Given the visual hints in this section that bunching may be a more salient issue in the German case, the large threshold effect found in this naive analysis may be upwardly biased (see Section 3.1). Therefore, I shed additional light on potential bunching effects and bunching behavior in the following section.

²⁷ For the choice of bandwidth in this specification, I use the estimated optimal bandwidths in column (4) as my benchmark and round up to the nearest whole number. Thus, I gain predefined and uniform bandwidths that I can use to calculate the bunching effects. Uniform bandwidths that encompass a fixed number of firms are important for calculating the lower bound on the threshold effect (see Section 5.4).

		40-Employee Threshold				
	(1)	(2)	(3)	(4)	(5)	
Effect (β_1)	0.345***	0.373***	0.386***	0.394***	0.388***	
Robust CI (1)	[0.318; 0.429]	[0.318; 0.456]	[0.320; 0.469]	[0.304; 0.492]	[0.185; 0.668]	
Bandwidth h	2.20; 3.38	4.15; 5.51	6.54; 7.58	8.31; 9.46	8;9	
Polynomial Order p	1	2	3	4	4	
Covariates Included	yes	yes	yes	yes	yes	
# of Observations	76,271	129,228	182,727	238,306	210,306	

Table 2: Threshold Effects (Dep. Var.: Number of Disabled Workers)

Notes: This table presents the estimation results for the effect of the threshold on the number of disabled workers in a firm (threshold = firm size of 40 employees). Basic covariates include firm age, regional characteristics (federal state) and industry. The bandwidths in columns (1)-(4) reflect the MSE-optimal bandwidths calculated with the *rdrobust* command in Stata. As the running variable (firm size) is discrete, estimates are adjusted for mass points in the running variable. Standard errors are clustered at the firm level. The robust confidence interval for the main specification in column (5) with standard errors clustered at the firm level and discrete values for the running variable (firm size) is [0.296; 0.557]. *** denotes statistical significance at the 1% level.

Source: BsbM and BHP 2004–2011, own calculations.

5.3 Unintended Effect: Bunching Below

This section analyzes the potential bunching effect that results from firms purposely staying below the firm size threshold. The histogram shown in Figure 6 further indicates the importance of manipulation. It shows that firm size density drops at the threshold, indicating that manipulation may indeed be an issue. I also formally test for the presence of a discontinuity in the firm size distribution (Cattaneo/Jannson/Ma, 2020). The test results suggest that the null that there is no bunching should be rejected at the 1 percent level (see Table B.1 and Table B.2 in the Appendix).²⁸ To quantify the extent of the bunching, I calculate the share of firms in each firm size category and run local regressions around the threshold with the calculated firm size density as the outcome variable. I again use different polynomial orders (2, 3 and 4) to check whether the results are sensitive to functional form.²⁹

Table 3 shows the results from estimating the bunching effects. The coefficient from the model with a second-order polynomial is -1.305. The models incorporating a more flexible functional form suggest larger bunching effects. When using a very flexible functional form, i.e., with a polynomial of order 4 and a fixed bandwidth of h_{below} =8 and h_{above} =9 based on my baseline specification in Section 5.2, the bunching effect is -2.017. This means that approximately 2 percent of the firms around the threshold have manipulated their size.

²⁸ I also perform the manipulation test proposed by Frandsen (2017) in the context of RDDs with a discrete running variable. This test also indicates that there is systematic manipulation of the running variable.

²⁹ Note that it is not possible to estimate a linear specification (*p*=1) with the MSE-optimal bandwidth calculation in this case due to the very small number of observations.

Taken together, the evidence suggests firms indeed seem to manipulate their size due to the (higher) noncompliance fine that is imposed for firms with 40 or more employees. This suggests that the large threshold effect on the number of disabled workers identified in Section 5.2 is upwardly biased.

Figure 6: Firm Size Density



Note: Histogram indicating firm size density around the 40-employee threshold. *Source:* BsbM and BHP 2004–2011, own calculations.

Table 3: Bunching Effects (Dep. Var.: Firm Size Density)

	40-Employee Threshold			
	(1)	(2)	(3)	(4)
Bunching Effect	-1.305***	-1.454***	-2.012**	-2.017
Robust CI	[-2.198; -0.604]	[-2.587; -0.535]	[-4.080; -0.212]	[-5.521; 1.702]
Bandwidth h	6.38; 7.20	8.34; 10.07	7.96; 11.03	8;9
Polynomial Order p	2	3	4	4
# of Observations	14	19	19	16

Notes: This table shows estimation results for the effect of the 40-employee threshold on firm size density (in %). ** and *** denote statistical significance at the 5% and 1% levels. *Source:* BsbM and BHP 2004–2011, own calculations.

5.4 Bounding the Effect

This section assesses the upward bias in the naive threshold effect and provides a lower bound on the effect, again following the strategy used by Lalive/Wuellrich/Zweimüller (2013). For this, I refer to my baseline specification in which h_{below} =8, h_{above} =9 and p=4 for both the bunching and the threshold effects. The bunching effect of -2.017 identified in Section 5.3 informs us about the absolute number of bunching firms, suggesting that 2.017 percent of the 210,306 firms considered within the fixed bandwidth manipulate their employment levels. Hence, there are 2,121 (=(0.02017*210,306)/2) employment manipulators in total.³⁰ As both types of firms, i.e., *noncompliers* and *perfect compliers*, may be *B-Firms* according to Section 3.2, I expect firms of each type to bunch below the threshold.

To assess how many of the 2,121 bunching firms are *B-perfect compliers*, I restrict my sample to firms that employ at least one disabled worker and estimate the bunching and threshold effects for this subsample of 121,382 observations. The result is an estimated bunching effect of -1.413 and an estimated threshold effect of 0.262 (see also Figure B.1 and Table B.3 in the Appendix). This result suggests that 858 of the 2,121 bunching firms are *B-perfect compliers* and 1,263 firms are *B-noncompliers*.³¹

To bound the threshold effect, I hypothetically reassign all potential bunching firms from a firm size of 39 employees to a firm size 40 while keeping the number of disabled workers constant (i.e., a total of 1,263 firms would still employ zero disabled workers, and 858 firms would still employ one disabled worker). I then recalculate the raw difference in the mean number of disabled workers among firms with 39 employees and among those with 40 employees. This yields a difference of 0.161. The original raw difference in the mean number of disabled workers in those firms was 0.348, so the bias amounts to 0.348-0.161=0.187. Using this bias calculation to bound the naive threshold effect of 0.388 suggests that the lower bound of the effect is 0.201. Thus, even after taking potential bunching into account, I still obtain to a positive threshold effect. Taken together, my estimates suggest that the employment quota indeed induces firms to employ more disabled workers, but dependent on the extent of bunching, the real threshold effect may be considerably smaller than the naive effect.

³⁰ The following example illustrates why this number is divided by two: Imagine 100 firms on either side of the threshold. Now assume that ten firms bunch and purposely stay below the threshold. Now, there are 110 firms below and 90 firms above the threshold. The resulting difference in the number of firms is 20 – twice the number of bunching firms (Lalive/Wuellrich/Zweimüller, 2013).

³¹ As a robustness check, I restrict my sample to firms that employ at least two disabled workers. As these firms (*overcompliers*) do not face additional costs at the threshold when they are below the threshold, bunching should not occur. In fact, Figure B.2 in the Appendix suggests that *bunching below the threshold* is not relevant for *overcompliers*. According to the formal test by Cattaneo/Jannson/Ma (2020), firm size density *increases* significantly for *overcompliers* at the threshold of 40 employees. This is plausible, as it is in line with the institutional regulations under which firms above the threshold are obliged to employ two and more disabled workers. Thus, the share of these firms probably increases at the threshold.

5.5 Bunching Behavior

To shed additional light on the bunching behavior of firms, I use the characteristics of the firms' workforces, which may be affected by bunching, as unintended outcome variables. Specifically, I look examine firm and employee productivity, firm dynamics and workforce composition.

The graphical inspection of selected variables shown in Figures 7, 8 and 9 suggests that there are discontinuities at the threshold: Median wages are considerably lower in firms below the threshold. In addition to wages, I use firm and person fixed effects - also called AKM effects – as a proxy for the firm and employee productivity provided by Bellmann et al. (2020).³² The illustration of the firm fixed effects analysis in Figure B.3 in the Appendix is very similar to that for wages. For person fixed effects, the graphical inspection also indicates a substantial discontinuity at the threshold (see Figure B.4 in the Appendix).³³ Furthermore, the share of regularly employed workers in firms below the threshold is lower than in firms above the threshold, whereas the share of marginally employed workers is higher. Last, firms just below the threshold have considerably lower employment growth. Table 4 reports the estimated discontinuities in the considered variables at the 40-employee threshold, again with different specifications (p=1 and p=4). The pattern of results supports the hypothesis that some firms bunch below the threshold and adjust their workforce when facing an increase in labor costs. Specifically, firms below the threshold substitute regularly employed workers with marginally employed workers who do not count toward the calculation of firm size (see Section 3.1). The significant discontinuities in wages and in AKM firm and person fixed effects suggest that adjusting the workforce may lead to lower productivity among bunching firms. These discontinuities may also result from selection, as low-productivity firms may be more incentivized to bunch since they face (relatively) higher costs at the threshold (see also the discussion and analysis of the heterogeneous effects between low- and high-wage firms in Section 5.7.1). In summary, the overall picture suggests that the increase in labor costs due to the noncompliance fine at the threshold of 40 employees is highly correlated with firm dynamics, firm productivity and firm employment structures.

When distinguishing between *nonompliers*, *perfect compliers* and *overcompliers*, the results in Table 4 show that the significant coefficients are mainly driven by *noncompliers*.³⁴ For *overcompliers*, in contrast, the coefficients are not significantly different from zero for any of

³² Table A.3 in the Appendix explains the construction of the AKM effects in more detail.

³³ I use the person fixed effects for 2003-2010 provided by Bellmann et al. (2020) for individuals in the Integrated Employment Biographies (IEB) employed in firms sized 28-51 in 2010. Thus, I obtain a baseline sample of 1,479,831 individuals.

³⁴ The graphical illustrations for the different types of firms are shown in Figures B.5, B.8, B.9 and B.10 in the Appendix.

the alternative outcomes. Koller/Schnabel/Wagner (2006) analyzed employment growth for firms around the (former) threshold within the German disabled worker law in 1999 and 2000. In line with my results, those authors also found evidence of a significant and substantial decline in employment growth among firms below the second threshold that do not employ any disabled workers.³⁵ Taken together, the results suggest that bunching behavior is particularly pronounced among those firms just below the threshold that face the highest costs at the threshold, as discussed theoretically in Section 3.2.



Note: This graph plots the ln of median wages by firm size around the threshold of 40 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.

IAB-Discussion Paper 25|2022

³⁵ Koller/Schnabel/Wagner (2006) estimate a different model specification. Specifically, the authors use a probit model, with the probability of growth in t+1 as the dependent variable. Their coefficient of interest is the interaction between being just below the former threshold (i.e., having 24 employees) and employing fewer than two disabled workers (i.e., not complying with the law). The coefficient on this interaction is -1.336 and is significant at the 5 percent level. Through simulations, the authors quantify the decline in growth as approximately 22.9 percentage points. The probable best approximation to this specification is to estimate the threshold effect on employment growth for all noncomplying firms (i.e., *noncompliers* and *perfect compliers*). This estimation (with *p*=4 and the MSE-optimal bandwidth calculation) results in a coefficient of 0.211, which is significantly different from zero at the one percent level. Thus, the effect is very similar to that found by Koller/Schnabel/Wagner (2006).

Figure 8: Firm Dynamics



Note: This graph plots employment growth by firm size around the threshold of 40 employees. Employment growth is defined via a dummy variable that equals 1 if a firm has more employees (according to the BsbM) in t+1 than in t and 0 otherwise. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.

5.6 Robustness Checks: Placebos and Donuts

To assess the credibility of my results, I perform several robustness checks. My first test is the use of placebo thresholds. For this, I estimate the discontinuities in the number of disabled workers per firm at firm sizes where there should be no discontinuities. Figure 10 shows the estimated discontinuities for the specification with p=4 and an optimal bandwidth of firm sizes 28-51 (including the true threshold at a firm size of 40).³⁶ The pattern of estimates displays a clear-cut peak at the true threshold. For some placebo thresholds, e.g., firm sizes of 28, 41, 42, 45 or 46 employees, the 95 percent confidence interval does not include 0. This is in contrast to the graphical illustration in Figure 5, which suggests that there are no discontinuities at these firm sizes. Specifications with different

³⁶ Note that the German labor law has additional regulations with other thresholds, which may also be relevant for the employment of disabled workers. One example is the threshold of 30 employees with regard to insurance for continued payment ("Entgeltfortzahlungsversicherung"). In Germany, employees are entitled to sick pay that is paid by their employer during the first six weeks of an illness. Health insurance reimburses employers for some of these costs through the insurance for continued payment. This insurance is obligatory for employers who do not employ more than 30 employees. For an overview of the German regulations with thresholds, see Koller (2010).

Figure 9: Regular and Marginal Employment



Note: This graphs plot (A) the share of regularly employed workers and (B) the share of marginally employed workers by firm size around the threshold of 40 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.

Table 4: Bunching Behavior

	p = 1		p = 4	4	
Dependent Variable	Total	Total	Noncompliers	Perfect	Over-
				Compliers	compliers
	(1)	(2)	(3)	(4)	(5)
Sociodemographic Structure					
Share of Females	-0.003	-0.011	-0.006	-0.015	-0.012
Share of Germans	0.005***	0.012**	0.023**	0.004	-0.000
Employment Structure					
Median Wages (ln)	0.050***	0.088***	0.095***	0.048*	0.036
Firm Fixed (AKM) Effects	0.029***	0.050***	0.048***	0.036**	0.022
Person Fixed (AKM) Effects	0.039***	0.033***	0.049***	0.029	-0.005
Share of Regularly Employed Workers	0.014***	0.016***	0.019***	0.015	0.007
Share of Marginally Employed Workers	-0.015***	-0.024***	-0.037***	-0.015	-0.004
Share of Apprentices	0.000	0.002	0.009	-0.003	-0.000
Share of Full-Time Workers	0.008**	0.015	0.012	0.022*	0.020
Share of Part-Time Workers	0.003	0.008	0.014	-0.005	-0.005
Skill Structure					
Share of Low-Skilled Workers	-0.005***	-0.010	-0.016*	-0.009	0.002
Share of Medium-Skilled Workers	-0.003	0.006	0.017	0.004	-0.021
Share of High-Skilled Workers	0.009***	0.007	0.004	0.002	0.017
Firm Dynamics					
Employment Growth in t+1	0.071***	0.229***	0.353***	0.116*	0.033

Notes: This table shows the estimation results for the effect of the threshold of 40 employees on alternative outcome variables. *Noncompliers, perfect compliers* and *overcompliers* are firms below the threshold that employ zero, exactly one or at least two disabled worker(s), respectively. All estimations are estimated by using the MSE-optimal bandwidth for either side of the threshold. Standard errors are clustered at the firm level. *, ** and *** denote statistical significance at the 10%, 5% and 1% levels. *Source:* BsbM and BHP 2004–2011 (2004–2010 for firm fixed effects, 2010 for person fixed effects), own

Source: BsbM and BHP 2004–2011 (2004–2010 for firm fixed effects, 2010 for person fixed effects), own calculations.

polynomial orders show that although there are significant discontinuities at some placebo thresholds, the robustness of these estimations seems to be low: While the estimated discontinuity at the true threshold is positive and highly significant in all specifications, the significance of the coefficients for the placebo thresholds varies considerably depending on the specification. Furthermore, in terms of magnitude, the coefficient at the true threshold is substantially larger than the coefficients at the placebo thresholds in most cases (see Figures B.11, B.12 and B.13 in the Appendix). Taken together, the overall pattern confirms the credibility of the estimated discontinuity at the true threshold of 40.

Next, Figure 5 suggests that, in particular, firms with 39 and 40 employees violate the otherwise quite linear relationship between firm size and the mean number of disabled workers. Thus, as a further robustness check, I perform donut estimations and exclude those firms (and other combinations of firms near the threshold) and calculate the

Figure 10: Placebo Thresholds



Note: The graph shows the effects of the placebo thresholds on the mean number of disabled workers with p=4 and an MSE-optimal bandwidth on either side of the threshold (controlling for predetermined covariates). All thresholds except for the 40-employee threshold are placebo thresholds. The 95% confidence interval refers to the robust CI estimated with the *rdrobust* command in Stata. As the point estimates could be outside the robust CIs, only the interval boundaries are shown. Standard errors are clustered at the firm level. *Source:* BsbM and BHP 2004–2011, own calculations.

bunching and threshold effects again for this subsample. Table 5 shows the results. Note that I now use a linear specification, as the overall relationship between firm size and the number of disabled workers – when excluding the nonlinear developments near the threshold – appears to be linear. Let us first turn to the threshold effects. Compared to the coefficients from the baseline specifications, the coefficients from the donut estimations are smaller but still highly significant. This confirms the notion that part of the estimated naive threshold effect is biased by firms bunching below the threshold. Regarding bunching, the estimations show that the bunching effects are considerably smaller than in my baseline estimations, indicating that firm size manipulations occur mainly among firms located directly around the threshold.

Overall, the significant threshold effects estimated for the subsamples without firms near the threshold confirm my main results: Even though bunching is present, the regulation seems to positively affect the number of disabled workers in firms.

Table 5: Donut Estimations				
	Baseline Estimation	Donut Estimations: Excluding Firms of Firm S		ms of Firm Size
		39	39+40	38+39
	(1)	(2)	(3)	(4)
		Bunching E	ffects	
Coefficient	-2.017	-0.608**	-0.521*	-0.568
Robust CI	[-5.521; 1.702]	[-1.459; -0.080]	[-1.364; 0.052]	[-2.396; 0.992]
# of Observations	16	15	14	14
		Threshold E	ffects	
Coefficient	0.388***	0.203***	0.164***	0.172***
Robust CI	[0.185; 0.668]	[0.233; 0.349]	[0.174; 0.301]	[0.155; 0.366]
Polynomial Order p	4	1	1	1
Covariates Included	yes	yes	yes	yes
# of Observations	210,306	192,965	182,616	177,260

Table 5: Donut Estimations

Notes: This table shows the estimation results for the effect of the threshold on the number of disabled workers in a firm (threshold = firm size of 40 employees). The bandwidth for all estimations is h_{below} =8 and h_{above} =9. Basic covariates include firm age, regional characteristics (federal state) and industry. Standard errors are clustered at the firm level. *, **, *** denote statistical significance at the 10%, 5% and 1% levels. *Source:* BsbM and BHP 2004–2011, own calculations.

5.6.1 Further Robustness Tests

Next, I exclude public administration firms, as the share of firms in this sector differs below and above the threshold (see Section 4.3). The estimated coefficients are similar to those from the baseline estimation (see Table B.6 in the Appendix). Thus, I can conclude that public administration firms do not alter my basic results. Furthermore, I exclude firms that are identifiable as multiestablishment firms.³⁷ The results for the threshold and bunching effects and – more importantly – the results regarding bunching behavior are also robust to this exclusion (see Table B.7 and Table B.8 in the Appendix). Last, altering the specification, for example, by using different kernel weights or using a different bandwidth selector, does not alter my results, either.³⁸

³⁷ I can identify a firm as multiestablishment firms in the BsbM data as soon as it reports a disabled worker who is not working in the main establishment. In this manner, I exclude 1,782 firm-year observations (of firms with 20 to 59 employees). Note, however, that this exclusion is selective in the sense that I exclude only firms employing at least one disabled worker.

³⁸ Results not shown but available upon request.

5.7 Heterogeneous Effects

Based on the theoretical considerations described in Section 3, I now turn to a discussion of the potential heterogeneity in the bunching and treatment effects. Specifically, I differentiate between low- and high-wage firms and analyze different industries.

5.7.1 Low-Wage and High-Wage Firms

First, with regard to low- and high-wage firms, the share of the noncompliance fine τ relative to wages is substantially higher for low-wage firms than for high-wage firms: Among firms with 20-59 employees, the relative shares of the noncompliance fine are approximately 7.1 percent (τ_1) and 12.2 percent (τ_2) of wages among firms in the first quartile of the wage distribution (low-wage firms).³⁹ In contrast, these shares are only approximately 3.0 percent (τ_1) and 5.1 percent (τ_2) of wages among firms in the fourth quartile (high-wage firms). Thus, the (relative) importance of the noncompliance fine differs considerably between these groups of firms. For a given wage w and disabled worker productivity p, a relatively larger noncompliance fine τ leads to an increase in the number *B*-firms among firms for which p < w and thus a larger bunching effect among low-wage firms. Likewise, a relatively larger fine leads to an increase in the number of *G*-firms among firms for which p > w. As a consequence, I also expect a larger threshold effect among low-wage firms (see Section 3.1). For the empirical analysis, I group the firms based on quartiles of the wage distribution. The graphical analysis shown in Figure 11 suggests that the threshold effect among firms in the first quartile of the wage distribution is larger than that among firms in the fourth quartile of the wage distribution. The estimated threshold and bunching effects shown in Table 7 confirm this notion: The threshold effect is substantially larger among low-wage firms. The naive threshold effect among low-wage firms is 0.588, compared to 0.235 among high-wage firms, and the lower bound of this effect is 0.301 among low-wage firms, compared to 0.072 among high-wage firms. Furthermore, bunching is present among both types of firms but is also more pronounced among low-wage firms. In summary, these results support the hypothesis that the threshold and bunching effects are larger among low-wage firms.

5.7.2 Effects by Industry

I next estimate the bunching and threshold effects stratified by industry. The results displayed in Table B.7 show that the largest (naive) threshold effects are found in the industries agriculture/fishery, construction and traffic/communication industries. Bunching

³⁹ The wage distribution is based on the median value of gross daily wages for full-time employees.

Figure 11: Mean Number of Disabled Workers



Note: These graphs plot the mean number of disabled workers in (A) low-wage firms and (B) high-wage firms by firm size around the threshold of 40 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.

Table 6: Heterogeneous Effects by Firm Wages

	Observations	Threshold	Effect (TE)	Bunching Effect	Lower Bound
		p = 1	p = 4	p = 4	of TE
	(1)	(2)	(3)	(4)	(5)
		Firms in 1s	t Quartile of	Wage Distribution	
MSE-Optimal Bandwidth		0.473***	0.588***	-2.963**	
Fixed Bandwidth	61,462	0.307***	0.590***	-2.598*	0.310
		Firms in 2n	d Quartile of	Wage Distribution	
MSE-Optimal Bandwidth		0.369***	0.398***	-1.913**	
Fixed Bandwidth	57,386	0.273***	0.392	-2.126	0.199
		Firms in 3re	d Quartile of	Wage Distribution	
MSE-Optimal Bandwidth		0.307***	0.334***	-1.643**	
Fixed Bandwidth	59,242	0.236***	0.318	-1.519*	0.151
		Firms in 4tl	h Quartile of	Wage Distribution	
MSE-Optimal Bandwidth		0.178***	0.233***	-0.826	
Fixed Bandwidth	74,540	0.126***	0.235***	-1.099**	0.072

Notes: This table shows the estimation results for the threshold effects (dependent variable: mean number of disabled workers in a firm) around the 40-employee threshold and the bunching effects (dependent variable: firm size density in %) stratified by firms' median daily wage (quartiles). Basic covariates include firm age and regional characteristics (federal state). Standard errors are clustered at the firm level. *, ** and *** denote statistical significance at the 10%, 5% and 1% levels. The fixed bandwidth is the MSE-optimal bandwidth from the estimation with polynomial order p=4, rounded up to the nearest whole number. *Source:* BsbM and BHP 2004–2011, own calculations.

is particularly pronounced in construction, traffic/communication and the public sector. When bounding the threshold effects by following the bounding exercise in 5.4, the largest lower bounds emerge in the agriculture/fishery, construction and other services sectors. Industries may serve as a proxy for different workplace characteristics. First, assuming that wages are equal between disabled and nondisabled workers, as done in Section 3.1, is accurate mainly for firms in industries with high levels of collective bargaining coverage. However, the results displayed in Table B.7 do not indicate a clear pattern in terms of the level of collective bargaining coverage: On the one hand, there are no significant threshold effects among firms in energy/mining and banking/insurance - both sectors typically characterized by a relatively high level of collective bargaining coverage (Ellguth/Kohaut, 2004, 2012).⁴⁰ On the other hand, there are substantial threshold (and bunching) effects among firms both in industries with high collective bargaining coverage (such as construction and public administration) and in industries with low collective bargaining coverage (such as other services and traffic/communication). Second, industries may have different working conditions, in particular different shares of physically demanding tasks. As physical disabilities still account for the majority of disabilities (Statistisches Bundesamt,

⁴⁰ Note that collective bargaining coverage also depends on establishment size. According to Ellguth/Kohaut (2012), 53 (39) percent of West German establishments with 10 to 49 (50 to 199) employees were not covered by a collective bargaining agreement in 2011.

2011), the share of physically demanding tasks in an industry may serve as a proxy for the average productivity gap between disabled and nondisabled workers. Sectors in which the average productivity of a disabled worker differs substantially from that of a nondisabled worker (i.e., $p \ll P$) probably have a higher share of firms in which p < w (and thus a higher share of bunching (*B*-)firms). The large bunching effect in the construction sector (-2.562) is in line with these considerations, as this industry is characterized by a high share of physically demanding tasks (Kroll, 2011).

	Observations	Threshold	Effect (TE)	Bunching Effect	Lower Bound
		p=1	p=4	p=4	of TE
	(1)	(2)	(3)	(4)	(5)
			Agriculture/	Fishery	
MSE-Optimal Bandwidth		0.226*	0.591**	-1.194***	
Fixed Bandwidth	4,356	0.218*	0.647	-1.519	0.420
			Energy/M	ining	
MSE-Optimal Bandwidth		0.235	0.279	-1.554	
Fixed Bandwidth	2,392	0.209	0.298	-1.591	-
			Manufact	uring	
MSE-Optimal Bandwidth		0.331***	0.337**	-1.704**	
Fixed Bandwidth	63,118	0.228***	0.336	-1.654	0.137
			Construc	tion	
MSE-Optimal Bandwidth		0.255***	0.536**	-2.517*	
Fixed Bandwidth	18,336	0.229***	0.536	-2.562	0.413
			Wholes	ale	
MSE-Optimal Bandwidth		0.301***	0.372***	-1.821**	
Fixed Bandwidth	46,454	0.198***	0.372*	-1.724*	0.205
		Ti	raffic/Comm	unication	
MSE-Optimal Bandwidth		0.352***	0.506***	-2.785*	
Fixed Bandwidth	16,763	0.257***	0.513*	-2.476*	0.241
			Banking/Ins	surance	
MSE-Optimal Bandwidth		0.101	0.081	0.187	
Fixed Bandwidth	3,151	0.039	-0.009	1.072	-
			Other Ser	vices	
MSE-Optimal Bandwidth		0.324***	0.428***	-2.053**	
Fixed Bandwidth	36,172	0.278***	0.569*	-2.046	0.362
		Pul	olic Administ	tration (PA)	
MSE-Optimal Bandwidth		0.334***	0.396***	-1.191	
Fixed Bandwidth	37,661	0.271***	0.377	-1.258**	0.186
		P	ublic Sector	(w/o PA)	
MSE-Optimal Bandwidth		0.355***	0.371*	-2.903*	
Fixed Bandwidth	11,666	0.265***	0.349	-1.818*	0.110

Notes: This table shows the estimation results for the threshold effects (dependent variable: mean number of disabled workers in a firm) around the 40-employee threshold and the bunching effects (dependent variable: firm size density in %) stratified by industry. Basic covariates include firm age and regional characteristics (federal state). Standard errors are clustered at the firm level. *, ** and *** denote statistical significance at the 10%, 5% and 1% levels. The fixed bandwidth is the MSE-optimal bandwidth from the estimation with polynomial order *p*=4, rounded up to the nearest whole number. For "energy/mining" and "banking/ insurance" industries, no significant threshold effects were identified and thus no lower bounds were calculated.

5.8 Results for the 60-Employee Threshold

In this section, I check whether a similar pattern is visible at the third threshold of 60 employees. Firms with 40 to less than 60 employees must employ at least two disabled workers, while firms with 60 or more employees are obliged to employ at least three (=five percent) disabled workers. Note, however, that there are other threshold rules for the 60-employee threshold in the German labor law.⁴¹ Thus, the following analyses are primarily exploratory and serve as a robustness check for the results for the 40-employee threshold.

I restrict my sample to firms around the threshold of 60 employees. Regarding the intended effect, the graphical illustration again indicates a considerable discontinuity in the mean number of disabled workers between firms below and above this threshold (see Figure 12). The histogram for the firm size distribution suggests that bunching is also present at this threshold (see Figure B.14 and, for the results of the formal test, Table B.4 in the Appendix). Furthermore, the plots of and estimations for selected alternative outcome variables regarding the employment and wage structures as well as firm dynamics are similar to the patterns in those outcome variables near the 40-employee threshold (see Figures B.15, B.16, B.17 and Table B.5 in the Appendix). Table 8 gives an overview of the formally estimated bunching and threshold effects for the 60-employee threshold. All threshold effects estimated with MSE-optimal bandwidths are significantly different from zero at least at the 10 percent level. In terms of size, the threshold effects for the 60-employee threshold are larger than those for the 40-employee threshold, while the sizes of the bunching effects are similar. This result is consistent with the results of Lalive/Wuellrich/Zweimüller (2013), who also find larger effects at higher thresholds (albeit without evidence of bunching at higher thresholds). Repeating the bounding exercise described in Section 5.4 yields a lower bound of the threshold effect of 0.380.⁴² In sum, the analyses for the 60-employee threshold largely confirm the results obtained for the 40-employee threshold.

⁴¹ For example, according to the Protection Against Dismissal Act (*Kündigungsschutzgesetz*), an employer with 60 or more employees must report a layoff of 10 percent of the workforce or of more than 25 employees to the employment agency.

⁴² Note that there are four firm types around the threshold of 60 employees: *noncompliers* with employment of disabled workers D = 0, *undercompliers* with D = 1, *perfect compliers* with D = 2 and *overcompliers* with $D \ge 3$. I estimate the share of *B-undercompliers* and *B-perfect compliers* among all bunching firms by restricting the sample to firms with at least one or at least two disabled workers, respectively. As a result, I find 237 *B-perfect compliers*, 765 *B-undercompliers* and 278 *B-noncompliers* among the 1,281 total bunching firms.

Figure 12: Number of Disabled Workers



Note: This graph plots the average number of disabled workers by firm size around the threshold of 60 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4).

Source: BsbM and BHP 2004–2011, own calculations.

	0				
		60-Employe	e Threshold – Bur	ching Effects	
	(1)	(2)	(3)	(4)	(5)
Coefficient		-1.691**	-1.905*	-2.006*	-2.161
Robust CI		[-3.482; -0.213]	[-4.202; 0.076]	[-4.528; 0.238]	[-7.084; 3.381]
Bandwidth h		6.75; 8.79	8.67; 12.53	11.51; 15.53	8;14
Polynomial Order p		2	3	4	4
# of Observations		15	21	27	21
		60-Employee	e Threshold – Thre	eshold Effects	
	(1)	(2)	(3)	(4)	(5)
Coefficient	0.462***	0.498***	0.529***	0.653***	0.653**
Robust CI	[0.410; 0.574]	[0.417; 0.618]	[0.420; 0.671]	[0.451; 0.902]	[0.160; 1.178]
Bandwidth h	3.48; 3.58	5.61; 6.77	7.44; 9.74	7.84; 13.63	8;14
Polynomial Order p	1	2	3	4	4
Covariates Included	yes	yes	yes	yes	yes
# of Observations	43,523	73,050	101,907	118,537	118,537

Table 8: Threshold and Bunching Effects for the 60-Employee Threshold

Notes: This table shows the estimation results for the bunching effects (dependent variable: firm size density in %) and the threshold effects (dependent variable: mean number of disabled workers in a firm around the 60-employee threshold). Basic covariates include firm age, regional characteristics (federal state) and industry. Standard errors are clustered at the firm level. *, ** and *** denote statistical significance at the 10%, 5% and 1% levels. It is not possible to estimate the bunching effect in a linear specification (p=1) with the MSEoptimal bandwidth calculation due to the very small number of observations. *Source:* BsbM and BHP 2004–2011, own calculations.

6 Summary and Conclusions

In Germany, firms with 40 or more employees are obliged to employ one additional worker with a disability. This paper analyzes the intended and unintended effects of this German employment quota for workers with disabilities. The intended effect refers to the effect of this regulatory threshold on firm demand for workers with disabilities, whereas the unintended effect refers to potential bunching below the threshold. Thus, my paper extends the literature on the effects of an increase in labor costs resulting from a disabled worker quota system.

I use this sharp increase in labor costs and adopt a threshold design, which is closely related to an RDD, to estimate these effects. However, the threshold design accounts for the fact that the running variable – firm size, in my case – is endogenous. My results indicate that the employment quota promotes the employment of disabled workers in firms located around the threshold. A naive estimate of the *intended*, or threshold, effect (when ignoring the bunching) suggests that threshold firms employ on average 0.388 more disabled workers. When analyzing the *unintended*, or bunching, effect, the results show that firms do indeed manipulate their employment due to the increase in labor costs at the threshold. The existence of bunching violates the assumptions necessary for identifying the unbiased effect of the regulatory threshold. However, based on the estimates about the extent to which firms manipulate, I am able to provide a lower bound of 0.201, which is still positive, though considerably smaller. Thus, the German noncompliance fine does indeed increase compliance with the quota and promote the employment of disabled workers.

However, the quota also has unintended consequences that can be harmful to overall employment: Firms just below the threshold have a lower probability of increasing employment and a higher probability of substituting away from regularly employed workers. This is interesting, as previous research has found little evidence of firms bunching below the labor law thresholds in Germany. In view of the multitude of threshold regulations in German labor law, my findings shed new light on the relevance of such thresholds. Further research should therefore emphasize the evaluation of regulatory thresholds and firm adaptation to such regulations in other contexts.

Appendix

Appendix A: Definitions and Institutional Details

Table A.1: Calculation of Firm Size According to the Disabled Worker Law (§ 156 and § 157 SGB IX))
Apprentices (including special trainee positions for lawyers and teachers)	

	Apprentices (including special trainee positions for lawyers and teachers)			
Evoluted groups of workers	Individuals who work less than 18 hours per week			
Excluded groups of workers	Individuals with a temporary contract of fewer than eight weeks			
	Individuals whose employment is not primarily for pay			
	(e.g., individuals whose employment is primarily for rehabilitation)			
	Individuals participating in job creation schemes according to SGB III			
	Individuals who are elected to their job after continuous practice			
Tomporal dimension	The relevant measure for the firm size is the annual average			
remporar dimension	of the monthly number of positions.			
Calculation dotails	Fractions of 0.5 or more are rounded down to the nearest whole number			
Calculation details	for firms with 20 to 59 positions			
	Fractions of 0.5 or more are rounded up to the nearest whole number			
	for firms with 60 or more positions			

Source: § 156 and § 157 SGB IX, own illustration.

Table A.2: Additional Definitions Related to Firms/Establishments

Definitions of Firms and Establishments

Legal Definition of "Employer" (Firm) According to the Disabled Worker Law: Employers can be either a natural or a legal person under public or private law as well as a company of any kind. Consequently, all employees of the same employer are included, regardless of the number of establishments or other locations over which they are distributed.

Definition of "Establishment" in the Administrative Data: An establishment is a regionally and economically delimited unit in which employees work. An establishment may consist of one or more branch offices or workplaces belonging to one company (Schmucker et al., 2018).

Source: Schmucker et al., 2018, own illustration.

Figure A.1: Share of Individual Establishments



Notes: This graph shows the share of establishments that are independent companies or independent organizations without any other places of business. The survey is representative of all establishments in Germany (Ellguth/Kohaut/Möller, 2014).

Source: IAB Establishment Panel, 2004-2011, own calculations.

Table A.3: Person and Establishment Fixed Effects ("AKM Effects")

AKM-Effects

AKM person and establishment fixed effects stem from a wage decomposition pioneered by Abowd/Kramarz/Margolis (1999), implemented for Germany by Card/Heining/Kline (2013), and updated by Bellmann et al. (2020). These effects are derived from the following wage model:

$$log(wage_{it}) = \alpha_i + \Psi_{J(i,t)} + x'_{it}\beta + \epsilon_{it},$$

where the log daily wages for worker i are the sum of a time-invariant person effect α_i , a time-invariant establishment effect $\Psi_{J(i,t)}$ for the establishment at which worker i is employed at time t, plus time-varying worker characteristics $x'_{it}\beta$, which affect all workers' wages equally at all establishments, and an error component ϵ_{it} , which is assumed to be independent of the right-hand-side variables. The estimates for the person effect α_i capture time-invariant individual characteristics that are rewarded equally across employers. Likewise, the index $x'_{it}\beta$ is interpreted as measuring the time varying worker characteristics that affect the productivity of worker i in all jobs. In x_{it} , an unrestricted set of year dummies and of quadratic and cubic terms in age fully interacted with education is included. Last, the establishment effect $\Psi_{J(i,t)}$ is interpreted as a proxy for an establishment productivity, as this effect represents the proportional pay premium (or discount) that is paid by establishment j to all employees (i.e., all those with J(i, t) = j) (Bellmann et al., 2020: p. 7).

Source: Abowd/Kramarz/Margolis (1999), Card/Heining/Kline (2013) and Bellmann et al. 2020.

Appendix B: Further Analyses

Table B.1: Manipulation Test by Cattaneo/Jannson/Ma (2020)

Cattaneo et al. Manipulation Test

(see Cattaneo/Jansson/Xinwei, 2018; Cattaneo/Jannson/Ma, 2020)

This test is based on a local polynomial density estimator and uses robust bias correction coupled with variance adjustments. Specifically, the manipulation test statistics in *rddensity* (Cattaneo/Jansson/Xinwei, 2018) take the form

$$T_p(h) = \frac{\hat{f}_{+,p}(h) - \hat{f}_{-,p}(h)}{\hat{V}_p(h)} \qquad T_p(h) \sim \mathcal{N}(0,1),$$

where *h* is the bandwidth and *p*, the polynomial order. $\hat{f}_{+,p}(h)$ and $\hat{f}_{-,p}$ are the local polynomial density estimators, and $\hat{V}_p(h)$ represents the corresponding SE estimator.

In Table B.2 and Table B.4, I estimate a model with data-driven MSE-optimal bandwidth choices (h_{MSE}) , p = 2 and a triangular kernel weight. With $q \ge p + 1$, the manipulation test takes the form of an α -level test, with the null rejected if

$$T_q^2(h_{MSE,p}) > X_1^2(1-\alpha)$$

 $T_q^2(h_{MSE,p})$ gives an asymptotically valid distributional approximation of $q \ge p + 1$. Thus, the possible first-order bias in the statistic T_p^2 is removed by using a higher-order polynomial in the estimation of the densities and adjusting the SE formulas accordingly.

Source: Cattaneo/Jansson/Xinwei (2018) and Cattaneo/Jannson/Ma (2020).

Table B.2: Cattaneo et al. Estimator Test Statistics

	Т	P > T
Robust	-21.2550	0.000
# of Observations	625.6	64

Source: BsbM and BHP 2004-2011, own calculations.

Table B.3: Effects among Firms with at Least One Disabled Worker

	Bunching Effect	Threshold Effect
Coefficient	-1.413	0.262**
Robust Cl	[-3.818; 1.240]	[0.084; 0.645]
# of Observations	16	121,382

Notes: This table shows the estimation results for the threshold effects on the number of disabled workers in a firm only for firms which employ at least 1 disabled worker (h_{below} =8, h_{above} =9; p=4). Standard errors are clustered at the firm level. ** denotes statistical significance at the 5% level. *Source:* BsbM and BHP 2004–2011, own calculations.



Figure B.1: Firm Size Density for Firms with at Least One Disabled Worker

Notes: Histogram of firm size density for firms with at least one disabled worker around the threshold of 40 employees.

Source: BsbM and BHP 2004–2011, own calculations.



Figure B.2: Firm Size Density for Firms with at Least Two Disabled Workers

Notes: Histogram of firm size density for firms with at least two disabled workers around the threshold of 40 employees.

Figure B.3: Firm Productivity



Note: The graph plots the AKM firm fixed effects (see Table A.3) by firm size around the threshold of 40 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004-2010, own calculations.





Note: This graph plots the AKM person fixed effects (see Table A.3) by firm size around the threshold of 40 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth $h_{below}=12$ and $h_{above}=12$).

Source: BsbM and Integrated Employment Biographies 2010, own calculations.

Figure B.5: Median Wages: Noncompliers, Perfect Compliers and Overcompliers



Note: These graphs plot the ln of median wages by firm size around the threshold of 40 employees separately for *noncompliers*, *perfect compliers* and *overcompliers*. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.



Figure B.6: Firm Productivity: Noncompliers, Perfect Compliers and Overcompliers

Note: These graphs plot the AKM firm fixed effects by firm size around the threshold of 40 employees separately for *noncompliers*, *perfect compliers* and *overcompliers*. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004-2010, own calculations.



Figure B.7: Employee Productivity: Noncompliers, Perfect Compliers and Overcompliers

Note: These graphs plot the AKM person fixed effects by firm size around the threshold of 40 employees separately for *noncompliers*, *perfect compliers* and *overcompliers*. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =12 and h_{above} =12). *Source:* BsbM and Integrated Employment Biographies 2010, own calculations.

Figure B.8: Employment Growth: Noncompliers, Perfect Compliers and Overcompliers



Note: These graphs plot employment growth by firm size around the threshold of 40 employees separately for *noncompliers, perfect compliers* and *overcompliers*. Employment growth is defined as a dummy variable equal to 1 if a firm has more employees (according to the BsbM) in t+1 than in t and equal to 0 otherwise. The black line approximates the functional form of the running variable (here with polynomial order fit *p*=4 and bandwidth h_{below} =20 and h_{above} =19).

Source: BsbM and BHP 2004–2011, own calculations.

Figure B.9: Share of Regularly Employed Workers: Noncompliers, Perfect Compliers and Overcompliers



Note: These graphs plot the share of regularly employed workers by firm size around the threshold of 40 employees separately for *noncompliers*, *perfect compliers* and *overcompliers*. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19).

Source: BsbM and BHP 2004–2011, own calculations.





Note: These graphs plot the share of marginally employed workers by firm size around the threshold of 40 employees separately for *noncompliers*, *perfect compliers* and *overcompliers*. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19).

Figure B.11: Placebo Thresholds (Polynomial Order 1)



Note: This graph shows the effects of placebo thresholds on the mean number of disabled workers for polynomial order p=1 and an MSE-optimal bandwidth on either side of the threshold (estimated including predetermined covariates). All thresholds except the 40-employee threshold are placebo thresholds. The 95% confidence interval is the robust CI estimated with the *rdrobust* command in Stata. As the point estimates could fall outside the robust CIs, only the interval boundaries are shown. Standard errors are clustered at the firm level. For c=41 and c=42, there were not enough observations to perform MSE-optimal bandwidth calculations.

Source: BsbM and BHP 2004–2011, own calculations.

Figure B.12: Placebo Thresholds (Polynomial Order 2)



Note: This graph shows the effects of placebo thresholds on the mean number of disabled workers for polynomial order p=2 and an MSE-optimal bandwidth on either side of the threshold (estimated including predetermined covariates). All thresholds except the 40-employee threshold are placebo thresholds. The 95% confidence interval is the robust CI estimated with the *rdrobust* command in Stata. As the point estimates could fall outside the robust CIs, only the interval boundaries are shown. Standard errors are clustered at the firm level.

Figure B.13: Placebo Thresholds (Polynomial Order 3)



Note: This graph shows the effects of placebo thresholds on the mean number of disabled workers for polynomial order p=3 and an MSE-optimal bandwidth on either side of the threshold (estimated including predetermined covariates). All thresholds except the 40-employee threshold are placebo thresholds. The 95% confidence interval is the robust CI estimated with the *rdrobust* command in Stata. As the point estimates could fall outside the robust CIs, only the interval boundaries are shown. Standard errors are clustered at the firm level.

Source: BsbM and BHP 2004–2011, own calculations.

Table B.4: Cattaneo et al. Estimator Test Statistics - 60-Employee Threshold				
	Т	P > T		
Robust	-22.5877	0.000		
# of Observations	266,44	86		

Figure B.14: Firm Size Density at the 60-Employee Threshold



Note: Histogram of firm size density around the threshold of 60 employees. *Source:* BsbM and BHP 2004–2011, own calculations.



Figure B.15: Firm Dynamics

Note: This graph plots employment growth by firm size around the threshold of 60 employees. Employment growth is defined as a dummy variable equal to 1 if a firm has more employees (according to the BsbM) in t+1 than in t and equal to 0 otherwise. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.



Figure B.16: Median Wages

Note: This graph plots the ln of median wages by firm size around the threshold of 60 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.

Figure B.17: Regular and Marginal Employment



Note: These graphs plot (A) the share of regularly employed workers and (B) the share of marginally employed workers by firm size around the threshold of 60 employees. The black line approximates the functional form of the running variable (here with polynomial order fit p=4 and bandwidth h_{below} =20 and h_{above} =19). *Source:* BsbM and BHP 2004–2011, own calculations.

Table B.5: Bunching Behavior – 60-Employee Threshold							
	p = 1		p = 4				
	(1)	(2)	(3)	(4)	(5)	(6)	
Dependent Variable	Total	Total	Non-	Under-	Perfect	Over-	
			compliers	compliers	Compliers	compliers	
Sociodem. Structure							
Females	0.001	0.001	-0.003	-0.001	0.038	-0.019	
Germans	0.003*	0.000	0.014	-0.016	0.003	-0.012	
Employment Structure							
Median Wages (ln)	0.048***	0.050	0.090***	0.045	-0.029	0.010	
Regularly Employed	0.009***	0.012*	0.025	-0.007	0.039*	0.006	
Marginally Employed	-0.009***	-0.014**	-0.033*	-0.007	-0.019	0.002	
Apprentices	-0.001	0.000	-0.002	0.007	-0.003	-0.009	
Full-Time Workers	0.002	0.008	-0.005	-0.007	0.042	0.023	
Part-Time Workers	0.007***	0.002	0.026	0.005	-0.013	-0.018	
Skill Structure							
Low-Skilled Workers	-0.006***	-0.007	-0.009	0.000	-0.001	0.008	
Medium-Skilled Workers	-0.001	-0.001	0.001	0.005	0.003	-0.005	
High-Skilled Workers	0.010***	0.021	0.018	0.016	-0.012	-0.004	
Firm Dynamics							
Growth	0.035**	0.031	0.120	0.069*	0.029	-0.315*	

Notes: This table shows estimation results for the effects of the threshold of 60 employees on alternative outcome variables. Noncompliers, undercompliers, perfect compliers and overcompliers are firms below the threshold that employ zero, exactly one, exactly two or at least three disabled worker(s), respectively. All estimations are estimated by using the MSE-optimal bandwidth on either side of the threshold. Standard errors are clustered at the firm level. *, ** and *** denote statistical significance at the 10%, 5% and 1% levels. *Source:* BsbM and BHP 2004–2011, own calculations.

Table B.6: Robustness Test Excluding Public Administration Firms

	-					
	40-Employee Threshold – Bunching Effects					
	(1)	(2)	(3)	(4)	(5)	
Coefficient		-1.339***	-1.488***	-1.940**	-2.025	
Robust Cl		[-2.348; -0.561]	[-2.725; -0.492]	[-4.033; -0.177]	[-5.769; 1.814]	
Bandwidth <i>h</i>		6.23; 7.32	8.34; 9.79	8.07; 10.89	8;10	
Polynomial Order p		2	3	4	4	
# of Observations		14	18	19	17	
	40-Threshold – Threshold Effects					
	(1)	(2)	(3)	(4)	(5)	
Coefficient	0.341***	0.373***	0.385***	0.394***	0.387***	
Robust CI	[0.312; 0.428]	[0.314; 0.463]	[0.316; 0.473]	[0.299; 0.497]	[0.186; 0.667]	
Bandwidth <i>h</i>	2.21; 3.54	4.08; 5.91	6.54; 8.02	8.36; 10.13	8;10	
Polynomial Order p	1	2	3	4	4	
Covariates Included	yes	yes	yes	yes	yes	
# of Observations	65,102	110,242	163,744	210,444	218,977	
Lower Bound of Threshold Effect						

Notes: This table shows the estimation results for the bunching effects (dependent variable: firm size density in %) and the threshold effects (dependent variable: mean number of disabled workers in a firm around the 40employee threshold) without public administration firms. Basic covariates include firm age, regional characteristics (federal state) and industry. Standard errors are clustered at the firm level. ** and *** denote statistical significance at the 5% and 1%-level.

	0					
	40-Employee Threshold – Bunching Effects					
	(1)	(2)	(3)	(4)	(5)	
Coefficient		-1.307***	-1.455***	-2.013**	-2.018	
Robust CI		[-2.199; -0.607]	[-2.584; -0.538]	[-4.075; -0.221]	[-5.509; 1.725]	
Bandwidth h		6.37; 7.20	8.34; 10.08	7.95; 11.05	8;9	
Polynomial Order p		2	3	4	4	
# of Observations		14	19	19	16	
	40-Employee Threshold – Threshold Effects					
	(1)	(2)	(3)	(4)	(5)	
Coefficient	0.345***	0.371***	0.383***	0.390***	0.383***	
Robust CI	[0.316; 0.427]	[0.316; 0.452]	[0.317; 0.466]	[0.297; 0.489]	[0.190; 0.678]	
Bandwidth h	2.23; 3.36	4.19; 5.36	6.56; 7.61	8.24; 9.45	8;9	
Polynomial Order p	1	2	3	4	4	
Covariates included	yes	yes	yes	yes		
# of Observations	76,005	128,795	182,109	237,516	209,599	
Lower Bound of Threshold Effect						

Table B.7: Robustness Test Excluding Multiestablishment Firms I

Notes: This table shows the estimation results for the bunching effects (dependent variable: firm size density in %) and the threshold effects (dependent variable: mean number of disabled workers in a firm around the 40-employee threshold) without identifiable multiestablishment firms. Basic covariates include firm age, regional characteristics (federal state) and industry. Standard errors are clustered at the firm level. ** and *** denote statistical significance at the 5% and 1% levels.

Source: BsbM and BHP 2004–2011, own calculations.

Table B.8: Robustness Test Excluding Multiestablishment Firms II

	<u> </u>				
	(1)	(2)	(3)	(4)	(5)
	p = 1			p = 4	
Dependent Variable	Total	Total	Noncompliers	Perfect Compliers	Overcompliers
Sociodemographic Structure					
Share of Females	-0.003	-0.011	-0.006	-0.017	-0.011
Share of Germans	0.005***	0.012*	0.022**	0.005	-0.000
Employment Structure					
Median Wages (ln)	0.050***	0.085***	0.095***	0.045*	0.027
AKM Firm Fixed Effects	0.029***	0.050***	0.048***	0.037**	0.020
AKM Person Fixed Effects	0.039***	0.033***	0.048***	0.043	-0.004
Share of Regularly Employed	0.014***	0.016***	0.019***	0.016*	0.007
Share of Marginally Employed	-0.015***	-0.024***	-0.037***	-0.015*	-0.004
Share of Apprentices	0.000	0.002	0.009	-0.003	-0.001
Share of Full-Time Workers	0.008**	0.015	0.012	0.024*	0.018
Share of Part-Time Workers	0.003	0.007	0.014	-0.005	-0.005
Skill Structure					
Share of Low-Skilled Workers	-0.005***	-0.010	-0.015	-0.010	0.002
Share of Medium-Skilled Workers	-0.003	0.006	0.016	0.005	-0.021
Share of High-Skilled Workers	0.009***	0.007	0.003	0.003	0.018
Firm Dynamics					
Employment Growth in t+1	0.071***	0.228***	0.354***	0.118*	.0263

Notes: This table shows the estimation results for the effect of the threshold of 40 employees on alternative outcome variables. *Noncompliers, perfect compliers* and *overcompliers* are firms below the threshold that employ zero, exactly one or at least two disabled worker(s), respectively. All estimations are estimated by using the MSE-optimal bandwidth on either side of the threshold. Standard errors are clustered at the firm level. *, ** and *** denote statistical significance at the 10%, 5% and 1% levels.

Source: BsbM and BHP 2004–2011 (2004-2010 for AKM firm fixed effects), own calculations.

References

- Abowd, J. M.; Kramarz, F.; Margolis, D. N. (1999): High Wage Workers and High Wage Firms. In: Econometrica, Vol. 67, No. 2, p. 251–333.
- Acemoglu, Daron; Angrist, Joshua (2001): Consequences of Employment Protection? The Case of the Americans with Disabilities Act. In: Journal of Political Economy, Vol. 109, No. 5, p. 915–957.
- Barnay, Thomas; Duguet, Emmanuel; Le Clainche, Christine; Videau, Yann (2019): An Evaluation of the 1987 French Disabled Workers Act: Better Paying Than Hiring. In: The European Journal of Health Economics, Vol. 20, p. 597–610.
- Bauer, Thomas K.; Bender, Stefan; Bonin, Holger (2007): Dismissal Protection and Worker Flows in Small Establishment. In: Economica, Vol. 74(296), p. 804–821.
- Bauernschuster, Stefan (2013): Dismissal protection and small firms' hirings: evidence from a policy reform. In: Small Business Economics, Vol. 40(2), p. 293–307.
- Beegle, Kathleen; Stock, Wendy A. (2003): The Labor Market Effects of Disability Discrimination Laws. In: The Journal of Human Resources, Vol. 38, No. 4, p. 806–859.
- Bellmann, Lisa; Lochner, Ben; Seth, Stefan; Wolter, Stefanie (2020): AKM effects for German labour market data. FDZ method report 01/2020, Research Data Center of the Federal Employment Agency.
- Bundesarbeitsgemeinschaft der Integrationsämter und Hauptfürsorgestellen (2020): BIH Jahresbericht 2019 | 2020. Behinderung und Beruf und soziale Entschädigung. [BIH Annual Report 2019 | 2020. Disability and Occupation and Social Compensation.]. Tech. Rep..
- Calonico, Sebastian; Cattaneo, Matias D.; Farrell, Max H. (2020): Optimal bandwidth choice for robust bias-corrected inference in regression discontinuity designs. In: Econometrics Journal, Vol. 23, p. 192–210.
- Calonico, Sebastian; Cattaneo, Matias D.; Titiunik, Rocio (2014): Robust Nonparametric Confidence Intervals for Regression-Discontinuity Designs. In: Econometrica, Vol. 82, No. 6, p. 2295–2326.
- Cameron, A. Colin; Miller, Douglas L. (2015): A Practitioner's Guide to Cluster-Robust Inference. In: The Journal of Human Resources, Vol. 50, No. 2, p. 317–372.
- Card, D.; Heining, J.; Kline, P. (2013): Workplace Heterogeneity and the Rise of West German Wage Inequality. In: The Quarterly Journal of Economics, Vol. 128, No. 3, p. 9671015.

- Cattaneo, M.; Jansson, M.; Xinwei, Ma (2018): Manipulation Testing Based on Density Discontinuity. In: The Stata Journal, Vol. 18, No. 1, p. 234–261.
- Cattaneo, M.; Vazquez-Bare, G. (2016): The Choice of Neighborhood in Regression Discontinuity Designs. In: Observational Studies, Vol. 2, p. 134–146.
- Cattaneo, M. D.; Idrobo, N.; Titiunik, R. (2019): A Practical Introduction to Regression Discontinuity Designs: Foundations. Cambridge University Press.
- Cattaneo, Matias D.; Jannson, Michael; Ma, Xinwei (2020): Simple Local Polynomial Density Estimators. In: Journal of the American Statistical Association, Vol. 115, No. 531, p. 1449–1455.
- Cattaneo, Matthias D.; Irobo, Nicolas; Titiunik, R. (2018): A Practical Introduction to Regression Discontinuity Designs: Volume II. Cambridge University Press.
- Ellguth, Peter; Kohaut, Susanne (2012): Tarifbindung und betriebliche Interessenvertretung: Aktuelle Ergebnisse aus dem IAB-Betriebspanel 2011. In: WSI Mitteilungen, Vol. 4, p. 297–305.
- Ellguth, Peter; Kohaut, Susanne (2004): Tarifbindung und betriebliche Interessenvertretung: Ergebnisse des IAB-Betriebspanels 2003. In: WSI Mitteilungen, Vol. 8, p. 450–461.
- Ellguth, Peter; Kohaut, Susanne; Möller, Iris (2014): The IAB Establishment PanelMethodological Essentials and Data Quality. In: Journal for Labour Market Research, Vol. 47(1-2), p. 27–41.
- Federal Employment Agency (2021): Tabellen, Schwerbehinderte Menschen in Beschäftigung (Anzeigeverfahren SGB IX) [Tables, Severely Disabled People in Employment]. Tech. Rep., Nuremberg.
- Federal Employment Agency (2014): Arbeitsmarkt in Zahlen. Beschäftigungsstatistik 2011 [Labor Market in Figures. Employment Statistics 2011]. Tech. Rep., Nuremberg.
- Federal Statistical Office (2013): Statistik der schwerbehinderten Menschen 2011 [Statistics of the Severely Disabled People 2011]. Tech. Rep., Federal Statistical Office.
- Frandsen, B. (2017): Party Bias in Union Representation Elections: Testing for Manipulation in the Regression Discontinuity Design When the Running Variable is Discrete. In: Cattaneo, M. D.; Escanciano, J. C. (Eds.), Advances in Econometrics: Vol. 38 Regression Discontinuity Designs: Theory and Applications, Bingley, UK: Emerald, p. 29–72.

German Bundestag (2011): Drucksache 17/5815.

Hiesinger, Karolin; Kubis, Alexander (2022): Beschäftigung von Menschen mit Schwerbehinderungen: Betrieben liegen oftmals zu wenige passende Bewerbungen vor. [Employment of people with severe disabilities: Establishments often receive too few suitable applications.]. Tech. Rep., IAB Kurzbericht 11/2022.

- Hijzen, Alexander; Mondauto, Leopoldo; Scarpetta, Stefano (2017): The Impact of Employment Protection on Temporary Employment: Evidence From a Regression Discontinuity Design. In: Labour Economics, Vol. 46, p. 64–76.
- Koller, Lena (2010): Ökonomische Auswirkungen arbeits- und sozialrechtlicher Schwellenwerte. Peter-Lang-Verlag: Frankfurt a. Main.
- Koller, Lena; Schnabel, Claus; Wagner, Joachim (2006): Arbeitsrechtliche Schwellenwerte und betriebliche Arbeitsplatzdynamik: Eine empirische Untersuchung am Beispiel des Schwerbehindertengesetzes. In: Zeitschrift für Arbeitsmarktforschung, Vol. 39, p. 181–199.
- Kroll, L. E. (2011): Konstruktion und Validierung eines allgemeinen Index für die Arbeitsbelastung in beruflichen Tätigkeiten auf Basis von ISCO-88 und KldB-92. In: Methoden, Daten, Analysen, Vol. 5, No. 1, p. 63–90.
- Lalive, Rafael; Wuellrich, Jean-Philippe; Zweimüller, Josef (2013): Do Financial Incentives Affect Firms Demand for Disabled Workers? In: Journal of the European Economic Association, Vol. 11, No. 1, p. 25–58.
- Lechner, Michael; Vazquez-Alvarez, Rosalia (2011): The effect of disability on labour market outcomes in Germany. In: Applied Economics, Vol. 43, No. 4, p. 389–412.
- McCrary, Justin (2008): Manipulation of the running variable in the regression discontinuity design: A density test. In: Journal of Econometrics, Vol. 142, No. 2, p. 698–714.
- OECD (2010): Sickness, Disability and Work: Breaking the Barriers A Synthesis of Findings Across OECD Countries. Tech. Rep., Organisation for Economic Co-operation and Development.
- OECD (2003): Transforming Disability into Ability: Policies to Promote Work and Income Security for Disabled People. Tech. Rep., Organisation for Economic Co-operation and Development.
- Schmucker, Alexander; Eberle, Johanna; Ganzer, Andreas; Stegmaier, Jens; Umkehrer, Matthias (2018): Establishment History Panel 1975-2016. In: FDZ-Datenreport, Vol. 1.
- Statistisches Bundesamt (2011): Statistik der schwerbehinderten Menschen.
- Verick, Sher (2004): Do Financial Incentives Promote the Employment of the Disabled? In: IZA Discussion Papers, , No. 1256.
- Wagner, Joachim; Schnabel, Claus; Kölling, Arnd (2001): Threshold Values in German Labor Law and Job Dynamics in Small Firms: The Case of the Disability Law. In: IFO-Studien, Vol. 47, No. 1, p. 65–75.

List of Figures

Figure 1:	Employment Obligation and Monthly Noncompliance Fines in Germany	11
Figure 2:	No Quota	14
Figure 3:	Quota	16
Figure 4:	Types of B-Firms at T_2	18
Figure 5:	Number of Disabled Workers	25
Figure 6:	Firm Size Density	27
Figure 7:	Wages	30
Figure 8:	Firm Dynamics	31
Figure 9:	Regular and Marginal Employment	32
Figure 10:	Placebo Thresholds	34
Figure 11:	Mean Number of Disabled Workers	37
Figure 12:	Number of Disabled Workers	41
Figure A.1:	Share of Individual Establishments	44
Figure B.1:	Firm Size Density for Firms with at Least One Disabled Worker	46
Figure B.2:	Firm Size Density for Firms with at Least Two Disabled Workers	46
Figure B.3:	Firm Productivity	47
Figure B.4:	Employee Productivity	47
Figure B.5:	Median Wages: Noncompliers, Perfect Compliers and Overcompliers	48
Figure B.6:	Firm Productivity: Noncompliers, Perfect Compliers and Overcompliers	48
Figure B.7:	Employee Productivity: Noncompliers, Perfect Compliers and Overcompliers	;48
Figure B.8:	Employment Growth: Noncompliers, Perfect Compliers and Overcompliers.	49
Figure B.9:	Share of Regularly Employed Workers: Noncompliers, Perfect Compliers	
	and Overcompliers	49
Figure B.10:	Share of Marginally Employed Workers: Noncompliers, Perfect Compliers	
	and Overcompliers	49
Figure B.11:	Placebo Thresholds (Polynomial Order 1)	50
Figure B.12:	Placebo Thresholds (Polynomial Order 2)	50
Figure B.13:	Placebo Thresholds (Polynomial Order 3)	51
Figure B.14:	Firm Size Density at the 60-Employee Threshold	52
Figure B.15:	Firm Dynamics	52
Figure B.16:	Median Wages	53
Figure B.17:	Regular and Marginal Employment	54

List of Tables

Table 1:	Descriptive Statistics for Firm Characteristics	23
Table 2:	Threshold Effects (Dep. Var.: Number of Disabled Workers)	26
Table 3:	Bunching Effects (Dep. Var.: Firm Size Density)	27
Table 4:	Bunching Behavior	33
Table 5:	Donut Estimations	35
Table 6:	Heterogeneous Effects by Firm Wages	38
Table 7:	Heterogeneous Effects by Industry	39
Table 8:	Threshold and Bunching Effects for the 60-Employee Threshold	41
Table A.1:	Calculation of Firm Size According to the Disabled Worker Law (§ 156 and	
	§ 157 SGB IX)	43
Table A.2:	Additional Definitions Related to Firms/Establishments	43
Table A.3:	Person and Establishment Fixed Effects ("AKM Effects")	44
Table B.1:	Manipulation Test by Cattaneo/Jannson/Ma (2020)	45
Table B.2:	Cattaneo et al. Estimator Test Statistics	45
Table B.3:	Effects among Firms with at Least One Disabled Worker	45
Table B.4:	Cattaneo et al. Estimator Test Statistics - 60-Employee Threshold	51
Table B.5:	Bunching Behavior – 60-Employee Threshold	55
Table B.6:	Robustness Test Excluding Public Administration Firms	55
Table B.7:	Robustness Test Excluding Multiestablishment Firms I	56
Table B.8:	Robustness Test Excluding Multiestablishment Firms II	56

Imprint

IAB-Discussion Paper 25|2022

Publication Date

16 November 2022

Publisher

Institute for Employment Research of the Federal Employment Agency Regensburger Straße 104 90478 Nürnberg Germany

All rights reserved

This publication is published under the following Creative Commons licence: Attribution -ShareAlike 4.0 International (CC BY-SA 4.0) https://creativecommons.org/licenses/by-sa/4.0/deed.de

Download

https://doku.iab.de/discussionpapers/2022/dp2522.pdf

All publications in the series "IAB-Discusssion Paper" can be downloaded from

https://iab.de/en/publications/iab-publications/iab-discussion-paper-en/

Website

https://iab.de/en

ISSN

2195-2663

DOI

10.48720/IAB.DP.2225

Corresponding author

Karolin Hiesinger Phone: +49 911 179-8481 E-Mail: karolin.hiesinger@iab.de